

DEPARTMENT OF ECONOMICS AND BUSINESS ECONOMICS AARHUS UNIVERSITY



# A multilevel factor approach for the analysis of CDS commonality and risk contribution

Carlos Vladimir Rodríguez-Caballero and Massimiliano Caporin

**CREATES Research Paper 2018-33** 

Department of Economics and Business Economics Aarhus University Fuglesangs Allé 4 DK-8210 Aarhus V Denmark Email: oekonomi@au.dk Tel: +45 8716 5515

## A multilevel factor approach for the analysis of CDS commonality and risk contribution

Carlos Vladimir Rodríguez-Caballero\* Massimiliano Caporin<sup>†</sup>

December 11, 2018

### Abstract

We introduce a novel multilevel factor model that allows for the presence of global and pervasive factors, local factors and semi-pervasive factors, and that captures common features across subsets of the variables of interest. We develop a model estimation procedure and provide a simulation experiment addressing the consistency of our proposal. We complete the analyses by showing how our multilevel model might explain on the commonality across CDS premiums at the global level. In this respect, we cluster countries by either the Debt/GDP ratio or by sovereign ratings. We show that multilevel models are easier to interpret compared with factor models based on principal component analysis. Finally, we experiment how the multilevel model might allow the recovery of the risk contribution due to the latent factors within a basket of country CDS.

Keywords: multilevel factor models, risk contribution, CDS risk factors JEL Classification: C32, C38, E44, F30, G15.

## **1** Introduction

Common factors capture the most relevant time variation in that it is spread across the variables in a panel. Due to relevance and practicality when treating the problem of dimensionality in large data sets, factor models have been extensively used in finance and macroeconomics. For estimation and inferential theory under different frameworks, see among many others, Forni et al. (2000), Stock and Watson (2002), Bai and Ng (2002), Bai (2003), Forni et al. (2004), and Forni et al. (2005).

As a standard practice, most empirical studies rely on the existence of pervasive factors, that is, factors that spread over and explain the full cross-section of analysed variables. One of the most common approaches to recover factors is

<sup>\*</sup>Department of Statistics, ITAM and CREATES, Aarhus University. Email: vladimir.rodriguez@itam.mx. Departmento de Estadística. Instituto Tecnológico Autónomo de México. Río Hondo No.1, Col. Progreso Tizapán, Álvaro Obregón, CDMX. 01080. México.

<sup>&</sup>lt;sup>†</sup>Department of Statistical Sciences, University of Padova. Email: massimiliano.caporin@unipd.it

the principal component analysis (PCA), mainly in situations when N increases in contrast with a the state space setup that is used when N is small, Bai (2003). However, in recent years, multilevel factor structures have attracted attention in either theoretical or empirical research. The cornerstone of these models is to decompose the common factor structure into different levels, with factors associated to the full cross-section, i.e. pervasive, and factors that impact on and explain only a specific subgroup of variables, the non-pervasive factors. Such a possibility might become extremely relevant, as Boivin and Ng (2006) point out that factors that have stronger loadings on some specific groups of series than others may lead to biased or even inconsistent principal components estimates for the unobservable factors.

The literature includes different approaches leading to multilevel factor models. One of the possible choices implies a number of zero restrictions in the associated loadings matrix, as discussed in Wang (2010), Choi et al. (2018), and Breitung and Eickmeier (2016) for the stationary case, and Rodríguez-Caballero and Ergemen (2017) for the non-stationary case. In such proposals, the factor structure is split into pervasive and non-pervasive common factors that are usually identified as global and regional (or local) factors, as the application of these models usually split data on a geographical basis as in Kose et al. (2003). Global factors affect all variables in the panel, whereas regional ones affect only those in the respective region.

In this paper, we extend the literature of multilevel factor models by considering a new scenario in which the factor structure is more detailed with respect to the impact of factors on groups of variables. In this way, we are able to characterise both within-group and between-group variations. Thus, our factor structure allows for some interactions among groups or regions that do not contain information of the remaining groups. In turn, this allows disentangling the role of pervasive factors, the global ones, from the role of regional factors, active in a single local area or in a single group of variables, and, in the meantime, identifying factors associated with more than a single block. The latter factors are not pervasive in the cross-sectional dimensions, but allow for common behaviours across a subset of the blocks. We might refer to these interaction components as semi-pervasive factors.

The estimation method we propose for our novel multilevel factor model is similar in spirit to the sequential least square procedure proposed by Breitung and Eickmeier (2016). We employ a successive procedure of canonical correlation analysis (CCA) and PCA to obtain initial values for all the factors involved in the model structure. Then, a sequential least square procedure is used to estimate each one of the unobservable common factors. The different levels of factors are orthogonal to each other to ensure that the effects of a specific level of factors do not leak into other level. We also assess the methodology proposed in relatively small samples by Monte Carlo simulations through which we show that the method correctly identify the factors.

We accompany the methodological advancement with an empirical analysis

on the credit default swap (CDS) market. In particular, we analyse the multilevel factor structure that characterise a panel of sovereign CDS spreads for more than 50 countries with the purpose of identifying the sources of commonality and how these have an impact on the risk of a CDS portfolio. Our empirical study is thus related to the work of Longstaff et al. (2011), which also discusses the presence and sources of commonality across sovereign CDS spreads, but limits the analysis to the use of principal components. Our purpose is to show that the multilevel factor model might provide a different viewpoint by associating commonality to latent factors capturing different country features. In fact, the multilevel factor model builds on grouping criteria among the target variables, which are needed to disentangle the role of global factors from that of local factors, where the latter are group-specific, and further introduces semi-pervasive factors capturing across-groups patterns. In our case, we show that a multilevel model based on country groupings build upon the Debt/GDP ratio or the sovereign rating provide views that differ from those associated with principal components, and that are more easily economically interpreted. The analysis of commonality associated with economically based country groupings and the adoption of a multilevel model is new in the literature.

A further work linked to our analysis is that of Fabozzi et al. (2016), who employ PCA and independent component analysis over a collection of European sovereign spreads. Their purpose is to evaluate the evolution of risk in the CDS market building on the role of the latent factors. In our work, to further highlight the different views provided by multilevel models and principal components, we take a step in a direction close to that of Fabozzi et al. (2016), still focusing on the risk dimension, but from a different angle. In fact, we will show which is the role of principal components and latent factors extracted from a multilevel model in generating the risk of portfolios built with sovereign CDS. In this way, we highlight that multilevel models allow for a more detailed evaluation of the risk drivers, and show that grouping countries according to a given economic criteria has a crucial role. To the best of our knowledge, we are the first to take this analysis from a risk contribution perspective, using the tools introduced by Roncalli and Weisang (2016).

The analysis of different roles of global and local factors in the CDS spreads market is not novel, and has attracted considerable interest over the last decade. Longstaff et al. (2011) find a relevant role for global factors, as opposed to country-specific, or local, factors by using principal components. Ang and Longstaff (2013) decompose the CDS spread of European countries and U.S. states into systematic (i.e. common) and idiosyncratic (i.e. local) and use the decomposition to compare the two economic areas. Augustin and Tedongap (2016) build again on principal components to show the relevance of U.S.-based variables as global drivers of sovereign CDS spreads. Fender et al. (2012) adopt a different modelling framework for analysing the role of global and regional risk drivers. Overall, the use of principal components is widespread in studies dealing with the identification of CDS risk factors and in the analysis of commonality. A closely related study, though based on a different approach, is that of Kocsis and Monostori (2016), which uses

a hierarchical model to disentangle global and local factors. However, none of the previous studies allow for the existence of semi-pervasive factors as in our multilevel model. In our analyses, we contrast the multilevel outcomes with PCA and show that the two approaches lead to the identification of latent factors with a different exposure to macroeconomic and financial drivers. The different key is that the factors coming from PCA are more homogeneous in the exposure to economic drivers than the factors associated with country groupings and extracted from a multilevel model. This opens the door for the possibility of an easier identification and interpretation, from an economic viewpoint, of latent factors.

The remainder of the paper is organised as follows. Section 2 introduces the model and the estimation methodology used, while Section 3 presents a finite sample study based on Monte Carlo simulations. Section 4 describes the data and the approaches for country grouping, while Section 5 discusses the features of the multilevel-based factor and contrasts them with PCA. Section 6 presents the risk contribution analysis. Section 7 performs subsample analyses. Section 8 presents the conclusion of this paper.

## 2 A multilevel model with subgroup factors

We consider a block-structured factor model in which the unobservable common factor structure may be classified in many different levels according to the number of blocks formed by data. Our model differs from standard multilevel factor models recently used in the literature because we allow for interactions among blocks in contrast with the standard two-level approach that assumes only global and regional factors.

For clarity of exposition, consider a panel data formed by three different blocks of data,  $B_1$ ,  $B_2$ , and  $B_3$ . A general factor structure may be formed by the global factor,  $G_t$ , which is a top-level pervasive factor that affects all blocks of data, the pairwise factors,  $F_{kj,t} = (F'_{12,t}, F'_{13,t}, F'_{23,t})'$ , which are sublevel pervasive factors that affect only the blocks  $(B_1, B_2)$  in the case of  $F_{12,t}$ , for instance. Finally, the factor structure is also formed by the block-specific factors,  $F_{k,t} = (F'_{1,t}, F'_{2,t}, F'_{3,t})'$ , which are the non-pervasive factors that affect only a particular block. Such factor structure is displayed in the Venn diagram in Figure 1.

Then, the three-block factor model is written as

$$y_{k,it} = \gamma'_{k,i}G_t + \kappa'_{kj,i}F_{kj,t} + \lambda'_{k,i}F_{k,t} + u_{k,it},$$
(1)

where k = 1, 2, 3 indicates the block, index  $i = 1, ..., N_k$  denotes the i'th crosssection unit of block k, t = 1, ..., T is the time dimension, and kj means interaction between blocks k and  $j \in 1, 2, ..., k$  with  $k \neq j$ . The total number of cross-sectional units is  $N = N_1 + N_2 + N_3$ . Unobservable common factor structure is decomposed as discussed before: the  $\mathbf{r}_G \times 1$  vector  $G_t = (g_{1,t}, \ldots, g_{\mathbf{r}_G,t})'$  contains the  $\mathbf{r}_G$  unobservable global factor, the  $\mathbf{r}_{F_{kj}} \times 1$  vector



Figure 1: Factor structure formed by three different blocks of data.

 $F_{kj,t} = (f_{kj,1,t}, \dots, f_{kj,\mathbf{r}_{F_{kj}},t})'$  contain the pairwise block factors that interact only between blocks k and j with  $k \neq j$ , and the vector  $\mathbf{r}_{F_k} \times 1$  vector  $F_{k,t}$  consists of the  $\mathbf{r}_{F_k}$  unobservable block-specific factor of block k.  $\gamma_{k,i}, \kappa_{kj,i}$ , and  $\lambda_{k,i}$ are the  $\mathbf{r}_G, \mathbf{r}_{F_{kj}}$ , and  $\mathbf{r}_{F_k}$ - dimensional factor loadings. The number of global, pairwise, and block-specific factors as well as the cross-sectional dimension can naturally vary in each block k. The idiosyncratic term denoted by  $u_{k,it}$  satisfies, for our purposes, the standard assumptions of an approximate factor model, see Bai (2003) for the standard case or Wang (2010) and Choi et al. (2018) for the multilevel case. However, the model may also allow for long-range dependence processes from which the common component and the idiosyncratic components should follow the assumptions provided by Rodríguez-Caballero and Ergemen (2017).

We can rewrite (1) for the three blocks in a system way as

$$\begin{pmatrix} y_{1,\cdot t} \\ y_{2,\cdot t} \\ y_{3,\cdot t} \end{pmatrix} = \begin{pmatrix} \gamma_1 & \kappa_{12_1} & \kappa_{13_1} & 0 & \lambda_1 & 0 & 0 \\ \gamma_2 & \kappa_{12_2} & 0 & \kappa_{23_2} & 0 & \lambda_2 & 0 \\ \gamma_3 & 0 & \kappa_{13_3} & \kappa_{23_3} & 0 & 0 & \lambda_3 \end{pmatrix} \begin{pmatrix} G_t \\ F_{12,t} \\ F_{13,t} \\ F_{23,t} \\ F_{1,t} \\ F_{2,t} \\ F_{3,t} \end{pmatrix} + \begin{pmatrix} \epsilon_{1,\cdot t} \\ \epsilon_{2,\cdot t} \\ \epsilon_{3,\cdot t} \end{pmatrix} ,$$

$$y_t = \Lambda^* F_t^* + \epsilon_t, \tag{2}$$

where  $F_t^* = (G'_t, F'_{12,t}, F'_{13,t}, F'_{23,t}, F'_{1,t}, F'_{2,t}, F'_{3,t})'$  and  $\Lambda^* = [\Gamma_k, K_k, \Lambda_k]$ . The system is written in a matrix form as

$$Y = F^* \Lambda^{*'} + E_s$$

where the dimension of  $Y, F^*$ , and  $\Lambda^*$  being  $T \times N, T \times \mathbf{r_H}$ , and  $N \times \mathbf{r_H}$ , respectively, with  $\mathbf{r_H} = \mathbf{r}_G + \mathbf{r}_{F_{12}} + \mathbf{r}_{F_{13}} + \mathbf{r}_{F_{23}} + \mathbf{r}_{F_1} + \mathbf{r}_{F_2} + \mathbf{r}_{F_3}$  defining the total number of unobservable common factors.

Following ideas in Wang (2010), Choi et al. (2018), and Breitung and Eickmeier (2016), the factors loadings are identified up to a linear transformation of the following form,

$$A = \begin{pmatrix} A_{00} & 0 & 0 & 0 & 0 & 0 & 0 \\ A_{10} & A_{11} & 0 & 0 & A_{14} & A_{15} & A_{16} \\ A_{20} & 0 & A_{22} & 0 & A_{24} & A_{25} & A_{26} \\ A_{30} & 0 & 0 & A_{33} & A_{34} & A_{35} & A_{36} \\ A_{40} & A_{41} & A_{42} & A_{43} & A_{44} & 0 & 0 \\ A_{50} & A_{51} & A_{52} & A_{53} & 0 & A_{55} & 0 \\ A_{60} & A_{61} & A_{62} & A_{63} & 0 & 0 & A_{66} \end{pmatrix},$$
(3)

where to identify the factors, it is necessary to adapt usual normalisations as in PCA. First, orthonormal global, pairwise, and block-specific factors are given by  $A_{00} = \left(T^{-1}\sum_{t=1}^{T} G_t^0 G_t^{0'}\right)^{-1/2}$ ,  $A_{bb} = \left(T^{-1}\sum_{t=1}^{T} F_{kj,t}^0 F_{kj,t}^{0'}\right)^{-1/2}$  for b = 1, 2, 3 and  $k \neq j$ , and  $A_{bb} = \left(T^{-1}\sum_{t=1}^{T} F_{k,t}^0 F_{k,t}^{0'}\right)^{-1/2}$  with b = 4, 5, 6, respectively. Second,  $N^{-1}\Gamma'_k\Gamma_k$ ,  $N^{-1}K'_kK_k$ , and  $N^{-1}\Lambda'_k\Lambda_k$  are diagonal matrices. And third, the matrix A in (3) imposes that the blocks among global, pairwise, and block-specific factors are orthogonal with each other, implying that  $A_{b_kb_j} = 0 \forall b_k \neq b_j$  with  $b_k = (1 \dots, 6)$  and  $b_j = (0, 1 \dots, 6)$ .

The model specified by (1) can be extended to more than three blocks, although the complexity of the model as well as the number of restrictions for identification, as in (3) for the case of three blocks, will naturally increase with the number of blocks involved.

### 2.1 Estimation

The estimation procedure is based on the sequential approach proposed by Breitung and Eickmeier (2016) in which the main goal is to minimise the residual sums of square (RSS) function

$$S\left(F_{t}^{*},\Lambda^{*}\right) = \sum_{t=1}^{T} \left(y_{t} - \Lambda^{*}F_{t}^{*}(\hat{\theta})\right)'\left(y_{t} - \Lambda^{*}F_{t}^{*}\right)$$
(4)

by a sequence of two least-squares regressions until RSS achieves a minimum. The algorithm can be executed for the general case of k blocks as follow:

- 1. The algorithm starts by obtaining the initial values of the unobservable factors following the next strategy:
  - a) We emply canonical correlation analysis (CCA) on  $y_{k,it}$  to obtain the initial estimator of the global factor,  $\hat{G}^{(0)} = \left(\hat{G}_1^{(0)}, \hat{G}_2^{(0)}, \dots, \hat{G}_T^{(0)}\right)'$ .

- b) We regress  $y_{k,it}$  on  $\hat{G}_t^{(0)}$  in each block k to filter out the global component. Then, we get the residuals,  $y_{ki,t}^{*(0)}$ , from each of the k separate regressions.
- c) We again employ CCA on  $y_{k,it}^{*(0)}$  to obtain the next lower level block factors.
- d) Then, we regress  $y_{k,it}^{*(0)}$  on the respective block factors involved and get the residuals.
- e) Steps c) and d) are sequentially executed until the initial estimates of the pairwise block factors are obtained. We denote  $y_{k,it}^{**(0)}$  as the residuals after filtering the pairwise factors on each block k.
- f) Then, we run PCA on  $y_{k,it}^{**(0)}$  to get individual block factors  $\left(F_{1,it}^{(0)}, \ldots, F_{k,it}^{(0)}\right)'$ .
- g) Once initial estimators are obtained, the loading factors at the initial step are estimated from time-series regression of  $y_{ki,t}$  on the factors involved in each specific block k. Consequently, the factor loading matrix  $\hat{\Lambda}^{*(0)}$  is constructed.
- 2. The updated estimator for the unobservable common factors are obtained by a sequential procedure based on last reasoning and is executed as follow:
  - a) Run least-square  $y_{ki,t}$  on  $\hat{\Lambda}^{*(0)}$  to get  $G_t^{*(1)}$ .
  - b) Regress  $y_{k,it}$  on  $\hat{G}_t^{(1)}$  in each block k to filter out the global component. Get  $y_{k,it}^{*(1)}$ .
  - c) Run least-square  $y_{k,it}^{*(1)}$  on the next lower level block factors.
  - d) Repeat the same procedure as before until getting block-specific factors,  $\left(F_{1,it}^{(1)}, \dots, F_{k,it}^{(1)}\right)'$ .
  - e) Next, the updated (and normalised) factors  $F_t^{*(1)}$  are used to get the associated updated factor loading matrix  $\hat{\Lambda}^{*(1)}$ .
- 3. Step 2 is repeated until RSS converges to a minimum from which  $F_t^*$  and  $\hat{\Lambda}^*$ are collected.
- 4. The last step consists in orthogonalising each level of factor estimates with respect to the remaining factors. Such orthogonalisation can be sequentially executed as before. Since all factors are orthogonalised with each other, we can now perform a correct variance decomposition of individual variables with respect to each factor.

#### 3 **Monte Carlo Simulation**

In this section, using a Monte Carlo study, we examine the finite-sample properties of the estimation procedure proposed in Section 2.1. We focus on the case of three blocks for simplicity.

In our Monte Carlo study, the model in (1) is generated with three independent blocks, k = 3, with  $N_k \in \{20, 50, 200\}$  cross-sectional units in each block, and  $T \in \{150, 1500, 5000\}$  sample sizes. We consider for simplicity only one global factor,  $G_t$ , one factor in each pairwise block  $(F'_{12,t}, F'_{13,t}, F'_{23,t})'$ , and one block-specific factor in each block  $(F'_{1,t}, F'_{2,t}, F'_{3,t})'$ . All the unobservable factors are generated by a stationary AR(1) process where autoregressive parameters are taken as 0.5 with variances  $\sigma^2 \in \{1, 2\}$  to study the relative impact of a specific level of factor to the remaining ones. Furthermore, we consider that idiosyncratic terms  $u_{k,it} \stackrel{iid}{\sim} N(0, 2\phi)$  with  $\phi$  controlling the signal-to-noise ratio with  $\phi = \{5, 2, 0.5\}$ , corresponding to low, medium, and high signal-to-noise ratios. All loading factors are generated as N(1,1) following Boivin and Ng (2006). In each experiment, we regress the actual factors on the estimated ones to evaluate the reliability of the procedure and compute coefficient of determinations for the global, pairwise, and block-specific factors denoted as  $R_G^2, R_{F_{12}}^2, R_{F_{13}}^2, R_{F_{23}}^2, R_{F_1}^2, R_{F_2}^2$ . and  $R_{F_3}^2$ , respectively. These coefficients can be considered as a measure of consistency for all t, see Bai (2003). We compare the results obtained after applying the methodology proposed in this paper with the global and regional factors estimated by applying the methodology proposed by Breitung and Eickmeier (2016) in their two-level factor model. All simulations are based on 1000 replications of the model. Table 1 presents the results.

As can be seen from Table 1, the methodology proposed in this paper performs well in relatively small samples (Nk = 20, T = 150) and performs very well when sample sizes increase independently of size distortions between  $N_k$  and T. A low signal-to-noise ratio (rows with  $\phi = 5$ ) makes the factors independently of the level somewhat less precise, although such loss of precision is not dramatic. Furthermore, it seems that changes in the variances in the factors do not have a considerable impact in the estimation of the factors. Finally, we observe that in cases when a factor structure consists also of some pairwise factors, neglecting the existence of such factors, as in the two-level model proposed by Breitung and Eickmeier (2016), the estimation of global and regional factors will be substantially biased, performing very poorly even when the cross-section and time series dimensions considerably increase. In this sense, we can conclude that one should be cautious when analysing a panel data set consisting of different blocks of data.

## 4 Economic data and country grouping

The data set used in this paper collects the 5-year credit default swaps (CDS) premia on government bonds. We download the data from the Thomson Reuters Datastream database. Our dataset is a balanced panel consisting of 53 countries for each day for the period 1 January 2009 to 11 December 2017, yielding a total of 2,333 daily observations for each country. In our analyses, we work with the standardised log-changes of the CDS premia. Table 2 shows the countries included in the analysis as well as the descriptive statistics of the associated CDS logarithTable 1:  $T = 150, 1500, 5000, N_k \in (20, 50, 200)$ , and k = 3. The measure of consistency of the unobservable factors estimated are presented in the report.

					Mei	Methodology proposed in this paper	proposed	l in this p	tper		Breitu	Breitung and Eickmeier (2016)	ckmeier (	(2016)
φ	$\sigma_w$	$\sigma_{\nu}$	$\sigma_v$	$R_G^2$	$R_{F^2_{12}}$	$R_{F^{2}_{13}}$	$R_{F^{2}_{23}}$	$R_{F_1^2}$	$R_{F_2^2}$	$R_{F_3^2}$	$R_G^2$	$R_{F_1^2}$	$R_{F_2^2}$	$R_{F_3^2}$
								Nk =	20 $T$	$^{7} = 150$				
0.5	-	-	-	0.7977	0.9182	0.9317	0.8423	0.7684	0.8124	0.9493	0.6470	0.4183	0.3919	0.3811
0.5	0	1	1	0.8828	0.9448	0.9593	0.7845	0.9469	0.7763	0.8109	0.5289	0.8632	0.8712	0.8794
0.5	-	0	1	0.8154	0.8765	0.8717	0.8878	0.8748	0.9099	0.8515	0.0876	0.0986	0.0810	0.0846
0.5	1	1	0	0.9971	0.9393	0.9487	0.9465	0.9668	0.9534	0.9664	0.0711	0.3472	0.3436	0.3416
0	-	-	1	0.9225	0.7339	0.7705	0.7807	0.7820	0.8360	0.8771	0.6103	0.4064	0.3814	0.3665
0	0	-	1	0.8382	0.9218	0.7560	0.8405	0.7713	0.7729	0.7560	0.4964	0.8473	0.8628	0.8656
0	1	0	1	0.8557	0.8345	0.8171	0.8054	0.8518	0.9385	0.9162	0.0839	0.1041	0.0814	0.0828
0	-	-	0	0.9856	0.8362	0.8739	0.7686	0.8987	0.9082	0.8867	0.5651	0.3532	0.3385	0.3226
S	-	-	1	0.8846	0.7549	0.7463	0.6439	0.8484	0.7394	0.8304	0.5756	0.3922	0.3587	0.3678
S	0	1	1	0.7230	0.8941	0.7713	0.5382	0.6389	0.6449	0.6171	0.4758	0.8253	0.8360	0.8393
S	-	0	1	0.8808	0.7510	0.7386	0.7190	0.9354	0.9233	0.9361	0.0816	0.0964	0.0810	0.0786
S	1	1	0	0.9706	0.7269	0.7131	0.7235	0.8414	0.8198	0.8203	0.5471	0.3343	0.2987	0.3080

								$N_k = 5$	50 T	T = 1500				
0.5	1	1	1	0.9926	0.9855	0.9846	0.9819	0.9876	0.9894	0.9884	0.7583	0.4173	0.3896	0.3854
0.5	0	1	-	0.9291	0.9876	0.9850	0.8412	0.9660	0.9257	0.8593	0.5747	0.7072	0.6286	0.5256
0.5	1	0	1	0.9332	0.9705	0.9483	0.9641	0.9696	0.9441	0.9651	0.0512	0.0489	0.0705	0.0722
0.5	-	-	0	0.9988	0.9855	0.9845	0.9843	0.9905	0.9910	0.9899	0.4914	0.3347	0.3344	0.3416
0	1	1	1	0.9806	0.9549	0.9530	0.9507	0.9718	0.9716	0.9711	0.7169	0.4267	0.3964	0.4030
6	0	1	-	0.9106	0.9797	0.9762	0.8271	0.9375	0.9068	0.8394	0.5545	0.7003	0.6255	0.5254
2	1	0	1	0.9014	0.9410	0.9095	0.9282	0.9637	0.9252	0.9546	0.0537	0.0435	0.0705	0.0719
0	-	1	0	0.9957	0.9556	0.9564	0.9555	0.9739	0.9743	0.9744	0.4885	0.3431	0.3368	0.3317
5	-	1	-	0.9544	0.8974	0.8971	0.8916	0.9364	0.9367	0.9341	0.6809	0.4373	0.3924	0.3973
5	0	1	1	0.8787	0.9639	0.9581	0.7774	0.9016	0.8696	0.7784	0.5281	0.8866	0.4158	0.8153
5	1	0	1	0.9086	0.8884	0.8644	0.8864	0.9725	0.9368	0.9480	0.0541	0.0461	0.0714	0.0705
5	1	1	0	0.9894	0.9008	0.8992	0.8990	0.9408	0.9412	0.9413	0.4821	0.3269	0.3184	0.3302

(continued on next page).

	<b>~`</b>
	<u> </u>
	Ś
1	3
	2
	~
	~
1	~
- 1	0
. 1	<u> </u>
,	9
,	2
1/	<u></u>
1	5) 1
- 1 / -	6 I (C
1-1/-	
1-1-1/-	ole I (c
1-1-1/-	able 1 (c
7-1-1-1-1-	able 1 (c
T.L1, 1 / .	Table 1 (c

Methodology proposed in this paper

Breitung and Eickmeier (2016)

$\sigma_w   \sigma_ u   \sigma_v \left  \begin{array}{cc} R_G^2 & R_{F_{12}^2} \end{array}  ight $	$\sigma_v \mid R_G^2 = R_{I_1^2}^2$	$\mid R_{G}^{2}  R_{F_{12}^{2}}$			$R_{F^2_{13}}$	$R_{F^2_{23}}$	$R_{F_1^2}$	$R_{F_2^2}$	$R_{F_3^2}$	$R_G^2$	$R_{F_1^2}$	$R_{F_2^2}$	$R_{F_3^2}$
							$N_k = 2$	200 T	T = 5000				
1 1 1 0.9973 0.9959 0.9953	0.9959	0.9959	-	0.9953		0.9935	0.9957	0.9956	0.9966	0.8649	0.3844	0.3758	0.3786
-	0.9942	0.9942	-	0.9948		0.8633	0.9897	0.9501	0.8733	0.5949	0.7240	0.7442	0.7405
-	0.9766	0.9766	-	0.9452		0.9719	0.9725	0.9306	0.9568	0.0565	0.0105	0.0709	0.0704
-	0.9954	0.9954	-	0.9952		0.9943	0.9968	0.9956	0.9967	0.4981	0.3426	0.3369	0.3341
-	0.9884	0.9884	-	0.9879		0.9864	0.9908	0.9925	0.9915	0.8176	0.4482	0.4028	0.4067
-	0.9913	0.9913	-	0.9925		0.8485	0.9815	0.9388	0.8581	0.5751	0.9176	0.7470	0.5466
1 2 1 0.9262 0.9701 0.9346	0.9701	0.9701	-	0.9346		0.9719	0.9826	0.9271	0.9509	0.0546	0.0133	0.0724	0.0678
-	0.9884	0.9884	-	0.9888		0.9875	0.9932	0.9929	0.9934	0.4975	0.3407	0.3326	0.3387
-	0.9730	0.9730	-	0.9724		0.9696	0.9811	0.9832	0.9817	0.7417	0.4609	0.4211	0.4167
-	0.9879	0.9879	-	0.9898		0.8495	0.9772	0.9324	0.8525	0.5621	0.4121	0.4423	0.1450
1 2 1 0.8914 0.9499 0.9003	0.9499	0.9499	U	0.9003		0.9538	0.9789	0.8957	0.9359	0.0570	0.0128	0.0687	0.0698
1 1 2 0.9975 0.9742 0.9743	0.9742	0.9742	Ū	0.9743		0.9738	0.9852	0.9852	0.9854	0.4958	0.3282	0.3343	0.3340

*Notes:* The DGP is  $y_{k,it} = \gamma'_{k,i}G_t + \kappa'_{kj,i}F_{kj,t} + \lambda'_{k,i}F_{k,t} + u_{k,it}$ , where k = 1, 2, 3 and  $i \in (20, 50, 200)$  and  $T \in (150, 1500, 5000)$ .  $u_{k,it} \stackrel{iid}{\sim} N(0, 2\phi)$  are independently generated with  $\phi$  controlling the signal-to-noise-ratio with  $\phi = \{5, 2, 0.5\}$ . Only one top-level factor, one pairwise factor in each pairwise block, and one block-specific factor in each block are considered.  $G_t = 0.5G_{t-1} + w_t$  with  $w_t \sim IIDN(0, \sigma_w)$  and  $\sigma_w \in (1, 2), F_{kj,t} = 0.5F_{kj,t-1} + \nu_t$  with  $\nu_t \sim IIDN(0, \sigma_\nu)$  and  $\sigma_\nu \in (1, 2)$ , and  $F_{k,t} = 0.5F_{k,t-1} + v_t$  with  $v_t \sim IIDN(0, \sigma_v)$  and  $\sigma_v \in (1, 2)$ .  $R_G^2$  is the  $R^2$  of a regression of actual on estimates global factor. The same coefficient of determinations are computed according to the case. All experiments are based on 1000 replications.

Country	Min	Max	Skewness	Kurtosis	Country	Min	Max	Skewness	Kurtosis
Austria	-9.840	11.233	0.387	20.422	Kazakhstan	-6.022	5.683	0.286	8.279
Bahrain	-10.479	6.291	-0.397	14.812	Korea	-9.441	6.387	-0.043	10.226
Belgium	-14.511	6.786	-0.858	26.609	Latvia	-6.749	6.184	0.116	8.317
Brazil	-9.776	9.824	-0.064	27.767	Malaysia	-6.451	10.809	0.439	23.836
Chile	-14.644	12.920	0.243	65.317	Mexico	-5.531	6.176	0.155	6.461
China	-7.196	6.653	0.367	11.470	Netherland	-10.206	17.039	2.232	64.308
Colombia	-7.314	7.396	0.186	24.849	Norway	-7.190	7.830	-0.146	16.181
Costa Rica	-12.866	8.523	-0.604	28.990	Panama	-6.959	6.914	0.278	8.080
Cyprus	-7.888	10.879	0.348	15.648	Peru	-6.548	6.739	0.075	11.215
Czech	-6.915	10.675	0.568	14.861	Philipines	-14.104	11.871	-0.939	46.862
Denmark	-6.343	5.886	0.232	11.395	Poland	-6.602	5.970	0.168	7.105
Dominican	-27.370	8.471	-8.901	256.210	Portugal	-5.581	5.156	0.301	6.791
Dubai	-9.810	12.754	0.863	25.684	Qatar	-9.082	5.969	-0.031	13.237
Egypt	-9.054	8.653	0.257	13.291	Romania	-9.676	9.852	-0.594	21.566
El Salvador	-5.813	6.091	0.248	9.907	Russia	-15.302	17.492	0.693	85.444
Abu Dhabi	-7.998	11.497	1.126	34.920	Slovak	-7.703	7.275	0.183	8.394
Finland	-8.227	8.097	-0.030	14.803	Slovenia	-9.652	9.385	-0.575	41.021
France	-6.722	5.748	0.101	6.585	South africa	-6.451	9.529	0.752	18.187
Germany	-7.899	5.923	-0.139	12.017	Spain	-7.063	6.710	0.334	10.328
Guatemala	-9.709	11.930	1.161	33.471	Sweden	-12.008	6.677	-0.413	15.417
Hungary	-14.654	10.408	-3.884	74.420	Thailand	-7.2266	6.623	0.139	9.277
Iceland	-10.695	7.105	-0.859	23.754	Turkey	-10.337	9.019	-0.491	28.637
Indonesia	-7.642	11.245	0.569	16.293	UK	-6.275	6.124	0.184	7.346
Ireland	-10.126	9.466	0.002	25.936	USA	-9.351	7.262	-0.058	12.107
Israel	-8.524	10.200	0.270	19.944	Venezuela	-9.726	6.985	0.100	12.765
Italy	-6.446	8.351	0.330	8.414	Vietnam	-10.398	8.499	-0.474	20.639
Japan	-10.595	6.980	-0.139	17.565					

Ŀ.
sidered
<u>e</u>
· 2
ũ
8
untries c
<u>ě</u>
Ξ
п
б
S
53 co
0
the
Ę
series of
ŝ
Ξ
ē
Ss
~
9
-
_
. <u>N</u>
diz
lardiz
ndardiz
tandardiz
standare
ne standardiz
the standardiz
r the
for the
for the
tics for the
tics for the
tics for the
for the
ve statistics for the
tics for the
iptive statistics for the
iptive statistics for the
iptive statistics for the
escriptive statistics for the
: Descriptive statistics for the
2: Descriptive statistics for the
2: Descriptive statistics for the
2: Descriptive statistics for the
2: Descriptive statistics for the

mic returns. We stress that our dataset contains both developed countries as well as emerging countries. We define the panel composition in order to balance the cross-sectional dimension and the temporal coverage of the CDS data.

The adoption of a multilevel factor model requires the existence of groups, or clusters, among the variables of interest. We cluster the 53 countries in two different ways: i) using the median of the Debt/GDP ratio of at most 8 years (2009 - 2017), and ii) clustering the countries by the last credit rating assigned by Standard & Poor's. Given these choices for clustering countries, we will evaluate the role of both the credit rating and the Debt/GDP ratios in driving the commonality across the countries included in our panel. In fact, the multilevel model will provide, apart from a global factor, a set of latent factors associated with single country groups and across sets of country groups.

When clustering the countries with respect to the median of Debt/GDP ratio, we identify three different groups: the first comprises the 16 countries with the highest ratios, (i.e. over 70%); the second group includes the 19 countries with a Debt/GDP ratio between 70% and 45%; the last group contains the remaining 18 countries with the lowest ratio, (i.e below 45%). Table 3 illustrates the countries clustered by the level of Debt/GDP ratio, and the median ratios used to identify the groups.

We then cluster the countries with respect to the credit rating assigned by Standard & Poor's in 2017. The first group is formed by the 16 countries with the highest rating (above A+), the second group includes 10 countries, those with a high-medium rating (from A- to A+), the third is formed by the 14 countries with medium-low rating (from BBB+ to BBB-), and the remaining 13 countries are allocated in the lowest rating group. Table 4 depicts the countries clustered by the S&P credit rating.

The two grouping criteria are based on different, though linked, economic quantities. To verify if the two groups are somewhat related, we run a simple association analysis between the two country classifications, where classes correspond to groups; Table 5 contains the result. We observe how the two clustering criteria do not provide associated classifications. We read this evidence as reflecting the different informative content of the two clustering criteria. On the one hand, this suggests that by fitting a multilevel factor model on the two different country classifications, we could observe dissimilar results and recover interesting, though not aligned, economic interpretations. On the other hand, this might call for a more general clustering criteria, based on statistical clustering approaches. We also consider this additional possibility in a preliminary set of analyses. However, the findings, with respect to those obtained from multilevel models based on our economically based clustering, were clearly inferior. We thus decided to not report these additional evidences in the paper, but to made available them upon request. The differences between the two clustering criteria could also suggest to cross them to recover a finer classification (thus including 12 groups). Despite this being of some relevance, the multilevel model that we would define will be characterised

	Debt to	GDP Ratio (M	edian -2	2009-2017)	
Hig	,h	Mediur	n	Low	
Country	Ratio	Country	Ratio	Country	Ratio
Japan	186.27	Germany	68.30	Sweden	41.60
Italy	132.60	Colombia	67.28	Latvia	40.10
Portugal	130.40	Netherland	65.62	Panama	39.20
Slovak	112.00	Abu Dhabi	65.52	Korea	38.39
Cyprus	107.80	Dominican	64.87	Denmark	37.80
Belgium	105.90	Finland	63.60	Romania	37.60
Iceland	105.73	Vietnam	62.40	Czech	37.31
Ireland	100.43	Costa rica	62.00	Turkey	36.10
UK	96.78	Israel	61.90	Bahrain	35.94
France	96.00	Brazil	60.58	Norway	35.60
USA	94.41	Poland	54.10	Thailand	31.61
Egypt	92.30	Malaysia	52.16 Venezuela		28.20
Hungary	92.17	South africa			27.78
Spain	91.54	Philipines	51.00	Guatemala	24.27
Austria	84.60	El salvador	49.52	Peru	21.68
Slovenia	78.50	Mexico	47.90	Chile	21.30
		Dubai	47.60	Kazakhstan	11.74
		Qatar	47.60	Russia	9.10
		China	46.20		

Table 3: Countries clustered with respect to the median of Debt/GDP ratio

		Co	ountry ra	ting S&P 201	7		
Highe	st	High-M	edium	Medium-	Low	Lowes	t
Country	Rating	Country	Rating	Country	Rating	Country	Rating
Austria	AA+	Chile	A+	Colombia	BBB-	Brazil	BB-
Belgium	AA	China	A+	Hungary	BBB-	Costa rica	BB-
Czech	AA-	Iceland	А	Indonesia	BBB-	Cyprus	BB+
Denmark	AAA	Ireland	A+	Italy	BBB	Dominica	BB-
Dubai	AA	Israel	A+	Kazakhstan	BBB-	Guatemala	BB-
Abu Dhabi	AA	Japan	A+	Mexico	BBB+	Russia	BB+
Finland	AA+	Latvia	A-	Panama	BBB	South africa	BB
France	AA	Malaysia	A-	Peru	BBB+	Turkey	BB
Germany	AAA	Slovak	A+	Philipines	BBB	Vietnam	BB-
Korea	AA	Slovenia	A+	Poland	BBB+	Bahrain	B+
Netherland	AAA			Portugal	BBB-	Egypt	B-
Norway	AAA			Romania	BBB-	El Salvador	CCC+
Qatar	AA-			Spain	BBB+	Venezuela	SD
Sweden	AAA			Thailand	BBB+		
UK	AA						
USA	AA+						

Table 4: Countries clustered with respect to the last credit rating provided by S&P

by a very large number of potential factors (larger than the number of variables), requiring strong assumptions for identification.

We thus decided to not follow this line of research and maintain our focus on the two distinct classification criteria.

Rating / Debt/GDP	High	Medium	Low	All
High	5	6	5	16
Medium-High	5	3	2	10
Medium-Low	4	4	6	14
Low	2	6	5	13
All	16	19	18	53
Association test stati	stic = 3	3.973 - P-va	lue 0.6	8

Table 5: Frequency of countries over the two classification criteria - the last row reports the Chi-square test statistic for the null of no association between the two classification criteria for country grouping - the test statistic is distributed as a Chi-square with 6 degrees of freedom.

## 5 Global and economic-driven factors in CDS

We proceed with the estimation of the multilevel factor model on the CDS data. We consider two model specifications based on the two criteria used to classify the CDS data and described in the previous section. In detail, we fit a model for the Debt/GDP ratio classification and a model based on credit rating classification. Given that the two criteria lead to a different number of groups, the two fitted models will have a different number of factors.

In the following, in tables, figures as well as in the text, we will refer to groups by focusing either on the indicator level (Debt/GDP or rating) or on the factors. Table 6 shows the matching between the factors and the country groups. Overall, we have 15 factors in the rating case and 7 factors in the Debt/GDP case. The multilevel factor model for the Debt/GDP case exactly corresponds to the specification in equation 2, while the rating case has an equivalent, though more complex, structure. In the latter model, apart from the global factor and the group-specific, or local, factors, we have two collections of semi-pervasive factors: the first includes the four factors associated with the different sets of three country groups; and the second contains the six factors associated with pairs of groups.

In addition, for comparison purposes, we estimate the factor affecting the CDS evolution by PCA. To estimate the optimal number of unobservable factors, that is, the number of principal components to consider, we use the procedure proposed by Alessi et al. (2010) that improves the penalisation in the well-known criteria of Bai and Ng (2002). Such improvement is given by a tuning multiplicative constant that leads to heteroskedasticity robust inference. In our CDS dataset, we thus identify four factors, that is, we focus on the first four principal components.

		Rating		
Factor/Group	High	High-Medium	Medium-Low	Low
Global	X	X	Х	Х
F123	Х	Х	Х	
F124	Х	Х		Х
F134	Х		Х	Х
F234		Х	Х	Х
F12	Х	Х		
F13	Х		Х	
F14	Х			Х
F23		Х	Х	
F24		Х		Х
F34			Х	Х
F1	Х			
F2		Х		
F3			Х	
F4				Х
		Debt/GD	Р	
	High	Medium	Low	
Global	Х	Х	Х	
F12	Х	Х		
F13	Х		Х	
F23		Х	Х	
F1	Х			
F2		Х		
F3			Х	

Table 6: Matching among factors and country groups based on ratings or Debt/GDP ratio. Factors on the rows impact only on the marked groups (over columns).



Figure 2: Box plots of estimated coefficients (loadings) for country groups based on the Debt/GDP ratio (High, Medium and Low levels). By row, from top to bottom, left to right: loadings to the global factor; loadings to the HM factor; loadings to the ML factor; loadings to the HL factor; loadings to the single group factor.



Figure 3: Box plots of estimated coefficients (loadings) for country groups based on the sovereing debt rating (High, High-Medium, Medium-Low, and Low levels). By row, from top to bottom, left to right loadings to the factors: global; H-HM-ML, H-ML-L, HM-ML-L; H-HM; H-ML; H-L; HM-ML; HM-L; ML-L; single.



Figure 4: Box plots of loadings to the principal components from 1 to 4.



Figure 5: Box plots of estimated coefficients (loadings) to the principal components for country groups based on the Debt/GDP ratio (High, Medium and Low levels). By row, from top to bottom, left to right: loadings to the first factor; loadings to the second factor; loadings to the third factor; loadings to the fourth factor.



Figure 6: Box plots of estimated coefficients (loadings) to the principal components for country groups based on the sovereign debt rating (High, High-Medium, Medium-Low, and Low levels). By row, from top to bottom, left to right: loadings to the first factor; loadings to the second factor; loadings to the third factor; loadings to the fourth factor.

Figures 2 to 4 report the box-plots for the estimated loadings across country groups.<sup>1</sup> To avoid scale effects due to the variances of the factors, the figures report loadings scaled by the volatility of the factors. We observe that for the Debt/GDP case, the global factor appears to impact in a more relevant way to Medium and Low Debt/GDP ratio groups rather than for the High ratio group. Irrespective of the group, the global factor positively impacts on the CDS returns, allowing us to safely interpret it as a *market* factor. Factor F12, impacting only on High and Medium Debt/GDP ratio groups, is more relevant for the High group compared with the Medium. Notably, factors F13 and F23 (impacting on High-Low, and Medium-Low groups, respectively) seems to be less relevant as the median loadings are rather small. Factor F13 (High-Low) positively impacts on CDS in most cases. Finally, the group-specific factors (F1, F2 and F3) show average loadings

<sup>&</sup>lt;sup>1</sup>We chose to summarise loadings into box-plots given the large number of countries CDS we are considering. Detailed data, including the country-specific loadings, are available upon request.

close to zero for the Medium and Low Debt/GDP ratio groups (F2 and F3, respectively) and positive loadings to CDS for the High Debt/GDP ratio group. From an economic viewpoint, the countries with High Debt/GDP ratio are those less exposed to the global factor, which is interesting as it possibly signals the existence of relevant differences between this group and the other countries. We also note that the High Debt/GDP ratio group is also more exposed to the High-Medium factor (F12), which might thus be largely driven by the behaviour of the high group, and is positively exposed to the group-specific factor. This strengthens our view that the High Debt/GDP ratio country group is characterised by different behaviours in the CDS dynamic compared with the Medium and Low Debt/GDP groups.

Moving to the rating case, we observe that results are more heterogeneous. The global factor has positive loadings to CDS rates with larger impact for the two central groups (the High-Medium and Medium-Low rating groups). For combined factors (impacting on two or three rating groups), it is complex to identify patterns, even if, in some cases, we might observe a predominance of the combined factors for specific rating-based country groups. Finally, group-specific factors seem more relevant now, as they are characterised by loadings with larger sizes. These evidences might be due to the more complex structure of the multilevel model. Notably, both low rating and high rating groups seems to be less exposed to the global factor, possibly as a consequence of the heterogeneity within the groups. This is also in line with the relevant role played by the group-specific factors.

Finally, we consider the principal component analysis. Similarly to the multilevel factor models, we analyse the box-plots of the loadings, grouping them by either principal components, Figure 4, or coherently with the fitted multilevel models, Figures 6 and 5. We observe that the first principal component has the largest and more stable loadings across all the CDS, without remarkable differences across groups. Further, if we analyse the loadings of the components from the second to the fourth across the country groups (Debt/GDP ratio and rating), we do not see patterns that allow us to match principal components with groups. Apart from the first principal component, all other principal components have loadings more dispersed. This signals that the principal components have a less clear connection with the country groups based either on the Debt/GDP ratio or the sovereign rating, while the multilevel model, given it is grounded on an economically based classification criteria, provide interpretable factors by construction.

Apart from the evaluation of the loading, a first comparison between the two multilevel factor models and the more traditional PCA, might build on the ability of the models in capturing the behaviour of the CDS returns. We monitor this aspect by focusing on the fraction of the variance explained by the estimated factors, Figure 7, and the correlation between model residuals, Figure 8. Similarly to the loadings case, we compare PCA results by grouping them according to the groups adopted in the multilevel models. The fraction of variance explained by the PCA and by the multilevel model in the Debt/GDP case are very close, while the model based on rating groups provides more interesting and heterogeneous results. This could be a consequence of the larger flexibility of the model, which is capable of

capturing the behaviour of specific country groups. Notably, the extreme groups (High Debt/GDP ratio, Low and High rating) show a smaller fraction of explained variance. Again, we link this to the heterogeneity existing within the groups. Moving to the correlation analysis, the principal components seem to provide slightly better results compared with the multilevel models; in addition, the two multilevel factor models provide residual correlations that are very close.

Summarising our findings, by moving from principal components to multilevel models, we note relevant changes in the loadings to the latent factors despite the overall fit of the model being substantially equivalent. On the one hand, this challenges the potential benefits of the multilevel model, an aspect we will discuss in the following subsection. On the other hand, this highlights that a somewhat more detailed model built on economically based grouping criteria is expected to provide a better economic description of the relation among target variables (i.e. the CDS spreads) compared to a purely data-driven model.



Figure 7: Box plots of the fraction of variance explained by the various methods. Top row, left plot, multilevel factor model with country rating groups, right plot, multilevel factor model with Debt/GDP groups. Bottom row, PCA model evaluated on the same groups of the top plots.



Figure 8: Box plots of the correlations between the original series and the residuals from the multilevel models (top row: from left to right, country rating and Debt/GDP groups respectively.) and the PCA model (bottom row).

## 5.1 Are principal components and latent factors different?

We now take a deeper look at the differences between the latent factors recovered by our approach and factors obtained by a more standard principal component analysis. We start with a simple comparison between the dominant factor within the principal component setting and the global factor from our two multilevel models. Figure 9 shows the scatterplots between them.

We clearly note a positive correlations in both cases, thus suggesting that the dominant factors in the various approaches are possibly capturing the same latent behaviour. Table 7 shows the correlation between the first four principal components and the latent factors based on the Debt/GDP ratio. Notably, the global factor, despite being highly correlated with the first principal component (correlation equals 0.86), has a significant, and negative, correlation with the second and third principal components, equal to -0.31 and -0.36, respectively. Furthermore, there is no one-to-one matching between the latent factors and the first four principal components, as there are, overall, 14 correlations that are above 0.2 (in absolute terms), 50% of the full set of correlations. Therefore, the latent factors seem to provide



Figure 9: Scatterplot among the first factor obtained from standard principal component analysis and the global factors from both ways of clustering.

a different view on the country CDS with respect to the descriptive elements we might extract from PCA. The evaluation of Table 8 confirms this finding. Again, the first principal component and the global factor are highly correlated, the correlation coefficient equals 0.87, and the global factor is also negatively correlated with the second and third principal components, as in the Debt/GDP ratio case. Furthermore, there are 17 correlation coefficients that are above 0.2 in absolute terms. In addition, the factor F1, specific to the group with High rating, is negatively correlated to all the first four principal components. We can thus state that the two multilevel factor models provide latent factors differing from PCA, despite showing some common behaviour. To clarify in this respect, we report in Table 9 the correlations between the latent factors obtained from the two models. Notably, apart from the expected high correlation between the global factors, equal to 0.95, the subgroup factors do not show a clear match, confirming our intuition that the two approaches for country groupings provide different information.

Both the principal component approach and our multilevel model allow for analysing the commonality among CDS spreads. Both methods highlight the role of common latent factors, in particular the global one. However, the use of a multilevel model attributes the commonality not just to global factors or to country group-specific factors, but also to semi-pervasive factors that thus capture an intermediate commonality aspect among the CDS spreads. From a different viewpoint, we might interpret these semi-pervasive factors as linked to the heterogeneity of the target variables. While the global factors capture the overall patterns and the group-specific factor are associated with the common components within groups, the semi-pervasive factors capture the part of the homogeneity associated with the groups. Furthermore, while the principal components can be attributed ex-post, in

Factor/PC	PC1	PC2	PC3	PC4
Global	0.863	-0.312	-0.361	0.159
F12	0.204	0.441	0.109	0.078
F13	0.158	0.054	0.085	-0.655
F23	-0.218	0.389	-0.872	-0.044
F1	0.129	0.183	0.227	0.204
F2	-0.112	0.075	0.037	0.680
F3	0.335	0.717	0.182	-0.027

a more relevant way, to groups of countries, the factors extracted from a multilevel model comes, by construction, from specific groups of countries. Consequently, the interpretation of the model outcomes is simpler and immediate.

Table 7: Correlation between principal components (on columns) and factors of a multilevel model based on Debt/GDP ratio (1 is High, 2 is Medium and 3 is Low).

To further analyse the differences between the factors extracted by PCA and those estimated with a multilevel model, we regress the factors on a set of worldrelated macroeconomic variables, all expressed in daily returns over the same sample period of our analyses. We recover all data from Thomson Reuters Datastream. Our first and second explanatory variables are the Dow Jones World Developed and the Dow Jones World Emerging equity indexes. They will allow tracking of the equity market impact on the sovereign CDS risk, separating the role of developed markets from that of emerging markets. The third regressor is the VIX index, a fear index that we use to proxy the global uncertainty in the equity markets. It captures the potential impact of financial market uncertainty on the sovereign bond risk. Then, we include the oil price, a proxy for the commodity risk, which might have an indirect impact on both the oil importing and oil exporting countries. The impact will be mediated by the real economic effects of oil price changes. The fifth variable we include is the U.S. nominal dollar broad index, a proxy for the currency risk, a further indirect driver of possible changes in the sovereign risk. Finally, the last four variables, all from FTSE, track the bond market<sup>2</sup>, a world composite total return index for sovereign bonds (all maturities) to track the bond market impact on the CDS spreads; the differential between the returns on the world total return index for bonds with 10 years maturity and the returns on the world total returns index for bonds with maturity from 1 to 3 years to track the impact of maturity risk on the CDS spreads; the differential between the returns on the world total return index for sovereign bonds with A rating and the returns on the world total returns index for bonds with AAA rating to monitor the sovereign credit spread role; the differential between the returns on the world total return index for big corporate

<sup>&</sup>lt;sup>2</sup>Contrary to the usual practice, all our bond-related variables are total return indexes and not redemption yields, as the latter, unfortunately, were not available to us.

Factor/PC	PC1	PC2	PC3	PC4
GLOBAL	0.870	-0.392	-0.213	-0.017
F123	0.338	0.697	-0.188	-0.122
F124	-0.013	0.132	-0.271	0.578
F134	-0.209	-0.172	-0.538	0.165
F234	-0.103	0.170	-0.688	-0.299
F12	-0.006	0.010	0.049	0.291
F13	0.002	-0.053	0.070	-0.029
F13	-0.018	-0.112	0.017	-0.123
F23	0.012	-0.050	-0.053	0.290
F24	0.040	-0.090	-0.036	-0.101
F34	0.041	-0.125	0.009	-0.015
F1	-0.207	-0.473	-0.203	-0.258
F2	0.110	0.039	0.126	0.020
F3	-0.039	0.117	0.060	-0.423
F4	0.122	-0.013	0.085	-0.040

Table 8: Correlation between principal components (on columns) and factors of a multilevel model based on country rating (1 is High, 2 is High-Medium, 3 is Medium-Low, and 4 is Low).

	Global	High-Med	High-Low	Med-Low	High	Medium	Low
Global	0.950	0.008	0.114	-0.174	-0.120	-0.103	-0.011
H-HM-ML	0.121	0.362	0.332	0.455	0.478	-0.046	0.417
H-HM-L	0.137	0.187	-0.571	0.265	0.170	0.243	-0.049
H-ML-L	0.089	-0.228	-0.042	0.446	-0.227	0.171	-0.255
HM-ML-L	0.063	-0.066	0.043	0.651	-0.407	-0.174	0.068
H-HM	0.048	-0.141	0.185	0.003	0.064	0.617	0.035
H-ML	-0.014	0.219	0.104	-0.049	0.108	-0.018	-0.222
H-L	-0.017	-0.132	-0.066	-0.030	0.047	-0.294	-0.031
HM-ML	0.099	-0.306	-0.171	0.019	0.108	0.271	0.109
HM-L	0.045	-0.302	-0.175	0.052	0.324	-0.431	0.032
ML-L	0.066	-0.111	-0.150	-0.090	-0.124	-0.133	0.036
Н	0.008	-0.190	0.357	0.155	0.276	-0.058	-0.717
HM	0.036	0.021	0.274	-0.132	-0.043	0.210	0.101
ML	-0.148	-0.119	0.329	0.004	-0.187	-0.141	0.204
L	0.074	-0.063	0.057	-0.067	0.254	-0.070	0.018

Table 9: Correlation between the factors of the two multilevel models. On columns the factors based on the Debt/GDP classification (High, Medium and Low) while on the rows the factors based on the country rating classification (High, Medium-High, Medium-Low and Low).

bonds with BBB rating and the returns on the world total returns index for big corporate bonds with AAA rating to monitor the corporate credit spread role. Tables 10 - 12 report the estimated coefficients.

The factors react in a significant way to several macroeconomic world drivers. However, finding specific patterns for the multilevel factors seems hard. The sovereign credit spread is more relevant, across factors, than the corporate credit spread. The term *spread* becomes more relevant in the rating classification case, a somewhat expected result, as the sensitivity to the maturity for a specific country might be linked to the rating for the country debt. Equity indexes are also relevant, with the Emerging index prevailing over the Developed index. Finally, the currency seems also to be of some relevance. Overall, the factors react to different subsets of macroeconomic drivers but without a clear economic intuition. We read this evidence as distinctive of the model that builds factors on the basis of specific country groupings, without a clear association with the global economic drivers we select. The case of principal components is similar, but now the factors react, at most, to the same set of drivers, in particular the second and third principal components, making it virtually impossible to read their exposures from an economic viewpoint. Summarising, we believe that the adoption of a multilevel model lead to factors that are at least partially associated with an economic intuition, the one behind grouping criteria.

## 6 A risk contribution analysis

To further explain the relation between the factors and the risk of sovereign CDS co-movements, we proceed to an analysis focusing on the risk dimension. In a related work, (Fabozzi et al., 2016) analyse the volatility of latent factors extracted from European CDS by means of PCA or by independent component analysis (ICA). Their purpose was to highlight the sources of risk and relate the risk changes to policies. In our case, we are interested in characterising the role played by the various latent factors, coming from the multilevel model, in explaining the risk of different country groups.

Therefore, we adopt a recent risk decomposition derived from the work of (Roncalli and Weisang, 2016). We start from a general linear factor model, where the covariance matrix of  $R_t$  equals

$$\Sigma_R = \mathbf{B}\Sigma_F \mathbf{B}' + \Sigma_\varepsilon,\tag{5}$$

and where  $\Sigma_R$  is the covariance of  $R_t$ , **B** is the matrix of factor loadings,  $\Sigma_F$  is the covariance matrix among the factors, and  $\Sigma_{\varepsilon}$  is the residual covariance. For both multilevel factor models and for PCA-based factors,  $\Sigma_F$  is a diagonal matrix.

Despite that the covariance decomposition in (5) allows analysing the role of each factor in explaining the variance of each element in  $R_t$ , we prefer to work at an aggregate level, that is, by focusing on portfolios. As our purpose is to identify the risk drivers of sovereign CDS when countries are clustered according to a specific

Table 10: Regression Results for Debt to GDP group

			Deper	Dependent variable:			
	Global	F12	F13	F23	F1	F2	F3
	(1)	(2)	(3)	(4)	(5)	(9)	(2)
Equity_D	20.898	-23.384***	-20.263**	54.723***	-0.228	0.288	7.746***
	(16.502)	(8.498)	(9.439)	(10.073)	(0.166)	(1.117)	(1.411)
Equity_E	-210.126***	16.271***	6.237	-56.928***	0.203	0.163	-2.214**
	(11.238)	(5.227)	(6.742)	(7.021)	(0.126)	(0.684)	(0.967)
VIX	1.241	-1.148	-0.896	-5.351***	0.045***	0.083	$0.534^{***}$
	(1.310)	(0.711)	(0.645)	(0.835)	(0.016)	(0.104)	(0.111)
Oil	7.447**	1.541	2.171	$14.456^{***}$	-0.030	-0.013	-0.922***
	(3.259)	(1.591)	(1.621)	(2.039)	(0.033)	(0.239)	(0.350)
Uscurr	$104.421^{***}$	5.946	-14.386	-251.990***	-0.983**	-0.367	9.435***
	(39.881)	(19.552)	(23.113)	(26.157)	(0.485)	(2.686)	(3.651)
Bond_W	72.123**	-0.063	-0.990	-140.146***	-0.147	-4.015*	15.890***
	(34.382)	(18.488)	(17.748)	(21.423)	(0.448)	(2.417)	(3.412)
Bond_W_10+_minus_1_3	-52.505	-10.274	-13.331	103.728***	-0.229	1.437	-9.320***
	(32.822)	(15.591)	(13.518)	(18.963)	(0.373)	(1.756)	(3.059)
Bond_W_A_minus_AAA	-112.932***	-62.631***	-45.858**	-5.555	-2.481***	-2.197*	15.781***
	(29.833)	(13.191)	(18.484)	(15.675)	(0.465)	(1.288)	(2.526)
Bond_W_C_BBB_minus_AAA	-103.149***	-3.715	23.077	31.763*	$0.862^{**}$	-2.550	4.060
	(38.861)	(17.162)	(24.253)	(19.071)	(0.434)	(2.070)	(3.430)
Constant	0.065 (0.058)	0.001 (0.029)	0.002 (0.028)	-0.005 (0.036)	-0.0001 (0.001)	0.0002 (0.004)	-0.002 (0.006)
Adjusted R <sup>2</sup>	0.440	0.030	0.016	0.269	0.080	0.004	0.087

corporate bond indexes of A rated and AAA rated bonds and BBB rated and AAA rated bonds. Robust Newey-West standard errors are reported in is the FTSE all maturities sovereign bond index, Bond\_W\_10+\_minus\_1\_3 is the differential in the yield to maturities of 10+ years and 1-3 years FTSE sovereign bond indexes, Bond\_W\_A\_minus\_AAA and Bond\_W\_C\_BBB\_minus\_AAA are the differentialsin yield to maturities between FTSE Notes: Equity\_E is the return of the Dow Jones World Developed equity index, Equity\_D is the return of the Dow Jones World Emerging equity index, VIX is the change in the VIX index, Oil is the return on the oil price, Uscurr refers to change in the US nominal dollar broad index, Bond\_W parentheses. \*p<0.1; \*\*p<0.05; \*\*\*p<0.01. Table 11: Regression Results for Rating countries group

							Depender	Dependent variable:							
	Global	F123	F124	F134	F234	F12	F13	F14	F23	3	F24	F34	F1	F2	F3
	(1)	(2)	(3)	(4)	(5)	(9)	(7)	(8)	(6)	(10)	(11)	(12)	(13)	(14)	(15)
Equity_D	-5.830	-91.239*	3.446	38.407***	37.892***	3.687	6.688	-15.508**	4.915	4.915	3.406	-0.944***	390.978*	-88.832***	-98.188
	(16.726)	(52.233)	(7.751)	(8.951)	(8.133)	(6.498)	(4.179)	(6.472)	(6.313)	(6.313)	(6.388)	(0.232)	(230.987)	(24.281)	(527.285)
Equity_E	-179.158***	-96.971***	-10.497**	-40.078***	-41.639***	-10.235**	-5.968**	3.894	-4.570	-4.570	-5.360	$0.343^{**}$	-281.362*	79.933***	-593.504*
	(11.247)	(36.802)	(5.318)	(6.589)	(5.905)	(4.184)	(2.991)	(4.114)	(3.598)	(3.598)	(4.312)	(0.159)	(151.945)	(17.320)	(343.619)
VIX	1.811	-16.485***	-0.802	-1.204*	-3.568***	-0.560	0.002	-0.734	0.600	0.600	0.257	-0.059***	-18.023	-6.188***	16.172
	(1.334)	(4.767)	(0.547)	(0.702)	(0.647)	(0.527)	(0.345)	(0.474)	( $0.483$ )	(0.483)	(0.528)	(0.020)	(18.650)	(2.103)	(41.363)
Oil	3.932 (3.137)	57.576*** (11.427)	1.493 (1.382)	3.554** (1.711)	7.819*** (1.670)	-1.187 (1.281)	0.623 (0.939)	2.544** (1.294)	-0.008 (1.163)	-0.008 (1.163)	-0.326 (1.301)	0.015 (0.050)	-8.709 (45.311)	3.749 (5.098)	-169.237 (103.559)
Uscurr	200.068***	-811.093***	-20.905	-79.281***	-118.464***	-1.189	-11.348	-1.060	2.345	2.345	23.307	-0.296	-27.730	-3.393	-1,222.255
	(40.964)	(141.772)	(18.819)	(23.417)	(21.980)	(14.393)	(10.607)	(14.551)	(13.906)	(13.906)	(17.624)	(0.640)	(586.882)	(66.806)	(1,255.639)
Bond_W	121.864***	-549.525***	8.710	-4.874	-101.802***	8.214	-15.489	3.616	-7.565	-7.565	1.258	-1.632***	-322.860	20.134	-256.974
	(33.584)	(120.683)	(14.826)	(20.019)	(21.648)	(13.257)	(9.748)	(13.897)	(12.005)	(12.005)	(16.207)	(0.580)	(535.195)	(61.446)	(1,098.261)
Bond_W_10+_minus_1_3	-87.801***	257.449**	4.713	25.537	69.920***	-9.218	24.594***	-6.701	12.015	12.015	4.749	$1.099^{**}$	425.125	34.320	596.850
	(32.731)	(101.569)	(12.806)	(15.756)	(17.232)	(11.053)	(8.026)	(13.286)	(10.106)	(10.106)	(13.473)	(0.520)	(420.943)	(49.077)	(887.798)
Bond_W_A_minus_AAA	-91.007***	-1,013.980***	2.735	86.610***	33.301***	-6.801	4.519	0.059	0.231	0.231	8.494	-0.183	-59.397	23.978	-1,099.201*
	(29.148)	(132.820)	(10.432)	(11.010)	(11.015)	(7.447)	(6.989)	(7.024)	(8.963)	(8.963)	(8.992)	(0.678)	(309.077)	(28.331)	(643.213)
Bond_W_C_BBB_minus_AAA	-111.546***	199.939	4.796	25.793	-24.647	24.581**	-12.485	18.271	-14.299	-14.299	7.766	-1.201*	547.216	53.727	175.991
	(41.435)	(135.490)	(14.829)	(16.878)	(15.167)	(11.883)	(8.268)	(15.227)	(10.980)	(10.980)	(14.463)	(0.637)	(450.221)	(63.077)	(1,017.433)
Constant	0.065 (0.057)	0.012 (0.202)	0.001 (0.023)	-0.001 (0.032)	0.003 (0.031)	-0.001 (0.021)	-0.001 (0.016)	0.002 (0.023)	0.001 (0.020)	0.001 (0.020)	-0.002 (0.025)	0.0003 ( $0.001$ )	-0.175 (0.780)	-0.009 (0.092)	0.169 (1.866)
Adjusted R <sup>2</sup>	0.450	0.135	0.009	0.124	0.149	0.006	0.007	0.004	0.003	0.003	0.003	0.027	0.004	0.017	0.010
Notes: Equity_E is the return of the Dow Jones World Developed equity index, Equity_D is the return of the Dow Jones World Emerging equity	return of th	ie Dow Jon	es World	Develop	ed equity i	ndex, Eq	luity_D i	s the retu	urn of the	e Dow J	ones Wo	orld Emé	erging eq	uity	

is the FTSE all maturities sovereign bond index, Bond\_W\_10+\_minus\_1\_3 is the differential in the yield to maturities of 10+ years and 1-3 years FTSE sovereign bond indexes, Bond\_W\_A\_minus\_AAA and Bond\_W\_C\_BBB\_minus\_AAA are the differentialsin yield to maturities between FTSE corporate bond indexes of A rated and AAA rated bonds and BBB rated and AAA rated bonds. Robust Newey-West standard errors are reported in index, VIX is the change in the VIX index, Oil is the return on the oil price, Uscurr refers to change in the US nominal dollar broad index, Bond\_W parentheses. \*p<0.1; \*\*p<0.05; \*\*\*p<0.01. ž

		Dependent	variable:	
	F1	F2	F3	F4
	(1)	(2)	(3)	(4)
Equity_D	-9.133***	-5.621***	-9.321***	1.201
	(2.376)	(1.479)	(1.411)	(1.293)
Equity_E	-20.730***	5.289***	13.051***	-1.750**
-1	(1.596)	(0.957)	(0.969)	(0.886)
VIX	0.110	-0.745***	0.518***	0.131
VIA	(0.181)	(0.121)	(0.112)	(0.103)
				()
Oil	0.915*	1.376***	-1.675***	-0.258
	(0.483)	(0.309)	(0.274)	(0.243)
Uscurr	20.115***	-23.681***	22.563***	2.020
	(5.946)	(3.684)	(3.506)	(3.194)
Bond_W	5.802	-20.177***	11.247***	4.353*
	(5.010)	(3.257)	(2.972)	(2.589)
	(5.010)	(3.237)	(2.972)	(2.30))
Bond_W_10+_minus_1_3	-8.753*	11.879***	-9.882***	-1.388
	(4.730)	(2.743)	(2.679)	(2.033)
Bond_W_A_minus_AAA	-38.923***	-18.132***	-5.394***	1.889
	(4.867)	(2.130)	(1.850)	(1.919)
	(4.007)	(2.150)	(1.050)	(1.919)
Bond_W_C_BBB_minus_AAA	-15.960***	2.834	0.599	-1.070
	(5.915)	(3.351)	(2.758)	(3.043)
Constant	0.010	-0.001	-0.001	0.0003
Constant	(0.008)	(0.001)	(0.001)	(0.0003)
	(0.000)	(0.003)	(0.003)	(0.004)
Adjusted R <sup>2</sup>	0.518	0.144	0.207	0.009

Table 12: Regression Results for PCA

Notes: Equity\_E is the return of the Dow Jones World Developed equity index, Equity\_D is the return of the Dow Jones World Emerging equity index, VIX is the change in the VIX index, Oil is the return on the oil price, Uscurr refers to change in the US nominal dollar broad index, Bond\_W is the FTSE all maturities sovereign bond index, Bond\_W\_10+\_minus\_1\_3 is the differential in the yield to maturities of 10+ years and 1-3 years FTSE sovereign bond indexes, Bond\_W\_A\_minus\_AAA and Bond\_W\_C\_BBB\_minus\_AAA are the differentials yield to maturities between FTSE corporate bond indexes of A rated and AAA rated bonds and BBB rated and AAA rated bonds. Robust Newey-West standard errors are reported in parentheses. \*p<0.1; \*\*p<0.05; \*\*\*p<0.01

criteria, we chose to analyse the risk decomposition for equally weighted portfolios formed by the CDS of a specific country group. In particular, we consider four equally weighted portfolios when clustering countries according to the rating and three portfolios when clustering countries on the basis of the Debt/GDP ratio.

To measure the role played by each latent risk factor, we follow the approach of Roncalli and Weisang (2016) and perform a risk contribution decomposition in the presence of risk factors.

Start from the linear factor model for k assets and using m < k factors

$$R_t = BF_t + \varepsilon_t. \tag{6}$$

Then, consider a portfolio with weights vector equal to  $\omega$ . The weights represent the portfolio exposure to the assets, and using the linear model we can recover the portfolio exposure to the factors,  $\delta$ , as

$$r_{t,p} = \omega' R_t = \omega' B F_t + \omega' \varepsilon = \delta' F_t + \eta_t.$$
<sup>(7)</sup>

The exposure to the assets and the exposure to the factors are thus related by  $\delta = B'\omega$ . Roncalli and Weisang (2016) suggest using a decomposition of  $\omega$ introduced by Meucci (2007)

$$\omega = \begin{bmatrix} B^+ \tilde{B}^+ \end{bmatrix} \begin{bmatrix} \delta \\ \tilde{\delta} \end{bmatrix},\tag{8}$$

where  $B^+$  is the Moore-Penrose inverse of B and  $\tilde{B}^+$  is any matrix that spans the left null space of  $B^+$ . This decomposition links the portfolio weights to the portfolio factor loadings  $\delta$  and to the loadings to a set of residual factors ( $\tilde{\delta}$ ). Using this decomposition, we recover a decomposition of the total risk for a given portfolio. We use the portfolio volatility as a risk measure and therefore set

$$\mathbf{R}\left(\omega\right) = \sqrt{\omega' \Sigma_R \omega}.\tag{9}$$

In the absence of risk factors, the total portfolio risk can be decomposed in the contribution coming from the different assets,  $\Re C_i$ , with i = 1, 2, ..., k, with the following equalities (or Euler equation)

$$\mathbf{R}(\omega) = \sum_{i=1}^{k} \mathbf{RC}_{i} = \sum_{i=1}^{k} \frac{\partial \mathbf{R}(\omega)}{\partial \omega_{i}} \omega_{i}.$$
 (10)

Consequently, the risk contribution is equal to the product of the weight of an asset in the portfolio times the marginal risk of that asset. Moreover, the sum of risk contributions equals total risk.

In the presence of risk factors, and using (8), we have  $\Re(\omega) = \Re(\delta, \tilde{\delta})$  and the Euler equation is still valid and becomes

$$\mathbf{R}(\omega) = \frac{\partial \mathbf{R}\left(\delta, \tilde{\delta}\right)}{\partial \delta} \delta + \frac{\partial \mathbf{R}\left(\delta, \tilde{\delta}\right)}{\partial \tilde{\delta}} \tilde{\delta}.$$
(11)

The first term on the right-hand side of (11) is our quantity of interest, that is, the contribution to the total risk that we can attribute to the factors. The second term represents a residual risk component. For the first term, we can write

$$\frac{\partial \mathbf{R}\left(\delta,\tilde{\delta}\right)}{\partial\delta}\delta = \delta' \frac{\partial \mathbf{R}\left(\delta,\tilde{\delta}\right)}{\partial\delta'} \\
= \delta' \frac{\partial \mathbf{R}\left(\omega\left(\delta,\tilde{\delta}\right)\right)}{\partial\delta} \\
= \delta' \frac{\partial \omega\left(\delta,\tilde{\delta}\right)}{\partial\delta} \frac{\partial \mathbf{R}\left(\omega\right)}{\partial\omega'} \\
= \delta' B^{+} \frac{\partial \mathbf{R}\left(\omega\right)}{\partial\omega'} \\
= \omega' B B^{+} \frac{\partial \mathbf{R}\left(\omega\right)}{\partial\omega'},$$
(12)

where we have used (8). Given this result, when the risk measure is the portfolio volatility, the risk contribution due to factors equals

$$\frac{\partial \mathbb{R}\left(\delta,\tilde{\delta}\right)}{\partial\delta}\delta = \frac{\omega'BB^{+}\Sigma_{R}\omega}{\sqrt{\omega'\Sigma_{R}\omega}} = \sum_{i=1}^{m} \left(\omega'B\right)_{i} \left(B^{+}\Sigma_{R}\omega\right)_{i} \frac{1}{\mathbb{R}\left(\omega\right)} = \sum_{i=1}^{m} \mathbb{RC}\left(F_{i}\right).$$
(13)

The risk decomposition for portfolio  $\omega$  in the presence of risk factors thus becomes

$$\mathbf{R}\left(\omega\right) = \sum_{i=1}^{m} \mathbf{RC}\left(F_{i}\right) + \nu, \tag{14}$$

where the last term is the risk that cannot be explained by factors. The previous quantities might also be expressed in relative terms by standardizing all of them by the portfolio total risk  $R(\omega)$ .

The decomposition of the portfolio risk into the risk contribution of factors and the residual risk allows identifying the role exerted by each factor, and measuring the overall relevance of the factors. If the residual risk accounts for a large fraction of the total portfolio risk, we might be facing a missing factor or we might read this as an evidence for inappropriate factor decomposition. This is particularly relevant from a portfolio point of view, where the residual risk should play a minor role due to diversification benefits. In our setting, as the final purpose is to evaluate the role of the latent factors in driving the risk of the CDS over country groups, we decide to report the risk decomposition for country portfolios. The portfolios we consider are formed by all the countries belonging to one of the three groups by Debt/GDP ratio or to all the countries included in one of the four groups by sovereign debt rating. In all cases, we build portfolios with equal weights in order to highlight the trend within groups and to avoid overweighting single countries (which is possible due to the heterogeneity present within groups in terms of specific indicators that could have been used to define different weighting schemes). Tables 13 and 14 collect the results over the two country classifications, while Table 15 show the risk contribution due to principal components across the country portfolios based on either rating or Debt/GDP.

In the rating case, we note that the global factor is the most relevant risk contributor for all groups, and in particular for countries in the groups High-Medium and Medium-Low, where the factor accounts for 68% and 77% of the total risk, respectively. The risk contribution of the global factor is relatively lower in the High rating group, where it explains 41% of the total risk. Trivariate factors do have a relatively high contribution to the risk for specific country portfolios. The factor associated with High, High-Medium and Medium-Low rating groups (F123) has a relevant role for the High country rating group, explaining 23% of the total risk, and a significant role for the other two country groups (12% and 9%, respecitvely). The factors F134 and F234 are relevant for the High (F134), Medium-Low and Low groups (both F134 and F234) but with fractions of risk explained smaller than those observed for F123. The factor F124 is not significant at all. Bivariate factors are in general negligible in terms of risk contribution, apart for the two bivariate factors associated with the High and Medium-High (F12) and the Medium-Low and Low (F34) groups. The former has a minor relevance in explaining the risk of the two higher rating groups, while the latter is crucial in the risk of the Low rating group, spiking at the 15% of the risk explained. Group-specific factors assume a relevant role, in particular for higher rating groups, where they explain 29% and 15% of the total risk, respectively. Finally, the residual risk is very low, indicating that, from a risk perspective, the identified factors capture most of the risk in the country groups based on rating. The only exception is the lower rating group where the residual risk is 5.8%. This is completely different form what we observe in Table 15 where the residual risk, that is, the fraction of risk not explained by the principal components, is even 10 times higher than the fraction of unexplained risk in the multilevel model case. From an economic point of view, the evidences we have match those in terms of loadings, as we have a confirmation of the central role due to global factors, the relevant role of factors associated with extreme groups, and the importance of group-specific factors.

When looking at the risk decomposition for the Debt/GDP case, we note that all factors have now some impact on the various country portfolios. Moreover, the residual risk is higher than in the rating case, even if it remains at sensibly lower levels compared with the PCA risk contribution. Again, these results are in line with the findings associated with the loadings, but also highlight that the introduction of a finer classification of countries (four groups in the rating case versus the three groups in terms of Debt/GDP) seems to be more flexible in capturing the heterogeneity that characterises the countries as well as the country groups.

A final comment points at the PCA risk decomposition, where the first principal component has a central role, and is, overall, more relevant than the global factors extracted from the multilevel model. The second to the fourth components explain a limited fraction of the total risk, and the residual risk is much higher than in the multilevel model cases. This further confirms that by resorting to PCA we tend to assign a predominant role to the first component, which has an unclear economic intuition, apart from being closer to a weighed average of the variables (in our case, we do have positive loadings to the first component). Differently, a multilevel model allows identifying a collection of potentially relevant latent factors, which can surely be matched with groups of the variables. Consequently, if grouping is based on an economic criterion, we can associate factors to an economic intuition.

Overall, in our case, the risk contribution analysis highlight the role of latent factors for monitoring the risk of country groupings, in particular those based on ratings. The role of the global factor remains predominant, but we also highlight a relevant contribution from semi-pervasive factors associated with specific sets of country groups. Therefore, the adoption of economically based country grouping criteria in combination with a multilevel model could provide relevant insights in the analysis of the risk drivers of CDS spreads.

## 7 The relevance of multilevel factors during and after financial crises

We complete our analyses by looking at subsample results. Our data starts in 2009 and ends in 2017. In the first part of the sample, our data covers the last months of the global financial crisis as well as the European sovereign crises. Both these events had a relevant impact on the sovereign CDS market, see for instance, among many others, Caporin et al. (2017) and Caporin et al. (2018) and the cited references therein. Therefore, we run the previous analyses over two periods. The first, starting in 2009 and ending in 2012, and the second covering the years from 2013 to 2017. We comment here on the main findings; Appendixes A and B include all figures and tables.

In terms of loadings to latent factors, we do observe some differences. If we consider the Debt/GDP country groupings, the loadings observed in 2009 - 2012 for the global factor are higher than those recorded in 2013 - 2017. A possible explanation is the increased heterogeneity in the fundamental countries that, in turn, leads to an increased heterogeneity in the CDS behaviour and a decrease in the response to movements in the global factor. We might see a confirmation of this view in the larger dispersion of the loadings to the group-specific factors we

	High	Med-High	Med-Low	Low
GLOBAL	41.0	68.1	77.0	61.0
F123	23.1	11.6	8.6	0.0
F124	0.1	0.6	0.0	0.4
F134	4.0	0.0	5.1	7.8
F234	0.0	0.3	6.1	1.5
F12	1.5	3.1	0.0	0.0
F13	0.0	0.0	0.3	0.0
F14	0.0	0.0	0.0	0.0
F23	0.0	0.5	0.6	0.0
F24	0.0	0.0	0.0	-0.1
F34	0.0	0.0	0.5	15.2
F1	29.2	0.0	0.0	0.0
F2	0.0	15.0	0.0	0.0
F3	0.0	0.0	0.2	0.0
F4	0.0	0.0	0.0	8.3
Residual	1.0	0.7	1.6	5.8

Table 13: Risk contribution of the latent factor to the total risk of equally weighted portfolios by country groups based on ratings.

	High	Med	Low
GLOBAL	36.82	80.03	76.04
F12	42.21	5.19	0.00
F13	6.46	0.00	3.28
F23	0.00	5.29	8.73
F1	8.83	0.00	0.00
F2	0.00	5.38	0.00
F3	0.00	0.00	8.88
Residual	5.68	4.11	3.06

Table 14: Risk contribution of the latent factor to the total risk of equally weighted portfolios by country groups based on Debt/GDP.
		Debt/GDP			Rat	ing	
	High	Medium	Low	High	Med-High	Med-Low	Low
PC1	78.33	87.01	93.67	89.93	86.59	84.45	83.80
PC2	9.63	3.53	0.59	1.47	0.29	7.30	2.96
PC3	0.02	0.99	0.14	0.08	1.59	0.06	0.04
PC4	1.47	0.00	0.36	0.70	2.76	0.15	1.70
Residual	10.55	8.47	5.24	7.82	8.77	8.04	11.49

Table 15: Risk contribution of the PCA model on the total risk of equally weighted portfolios by country groups based on Debt/GDP or on country rating.

note in 2013 - 2017 compared with 2009 - 2012. For the semi-pervasive factors, the most relevant change is for the Medium-Low factor, where the loadings tend to move from positive to negative. Going to the rating country grouping case, results are less clear, even if we note, in 2013 - 2017, a lowering in the loading levels for the global factor and the semi-pervasive factor that excludes the Low rating countries. Finally, for the principal component factors, the most recent period shows loadings to principal components that are less dispersed but confirm the findings of the full sample, with loadings to the first component around 1. When intersecting the principal component loadings with the country groupings, we do not observe relevant patters in the two subsamples.

The correlation of residuals from the various models as well as the correlation among latent factors is in line with the full-sample analysis: a relevant reduction in the correlation among residuals, with close results between the PCA and the multilevel models; a high correlation among global factors and between them and the first principal component; the absence of a matching between the other principal components and the latent factors as well as between latent factors (global excluded) from the two multilevel models.

More interesting findings emerge from the risk contribution analysis. In the rating case, we note a first relevant change in the role exerted by the factor F124 (it excludes the Medium-Low rating countries): in the first sample, it represents a relevant fraction of the risk contribution (up to 7% for the Medium-High group), while in the period 2013 - 2017, its contribution is close to zero, coherently with the full-sample case. We motivate this change in country ratings across time with possible effects on the group composition and the subsequent factor interpretation. This is also in line with the second relevant modification we observe, associated with the increased role of the group-specific factor for the High rating countries. The role of the factor increases in the sample 2013 - 2017, jumping to 50% of the risk contribution for the High rating countries. This signals a sort of separation effect induced by tension in the CDS market after the global financial crisis and the European sovereign crisis, with High rating countries being influenced mostly from their own shocks and less dependent on global and semi-pervasive factors. For the latter, for the High rating group, we observe a decrease of the risk contribution from 43% to 32% for the global factor and from 18% to 10% for factor F123 (excluding Low rating countries). Similar, but less clear effect, is observed fro the Medium-High rating group, where the group-specific factor shows an increase in the risk contribution from 4% to 23%. We observe similar patterns in the Debt/GDP grouping case. The global factor contribution to the risk decreases in a significant way for the High Debt/GDP group, moving from 45% in 2009 - 2012 to 29% in 2013 - 2017, but also for the other two groups, from 84% to 71% for the Medium group and from 80% to 67% for the Low Debt/GDP ratio group. Furthermore, the factor F12, focusing on the High and Medium groups, shows some decrease in the risk contribution for the High group (from 46% to 37%) but the other two semi-pervasive factors, and in particular the F23, record a relevant increase, as their risk contribution multiplies by three. Finally, the specific factor for the High group moves from a 4% contribution in 2009 - 2012 to a 18% contribution in 2013 - 2017, thus again suggesting a relevant role for shocks associated with the group members. Finally, we observe that the residual term, not attributed to some of the factors, sensibly increases in the most recent sample. This could signal, again, that the post-crisis period has a behaviour that partially differs from the 2009 - 2012 period. In the principal component case, the risk contribution analysis over the two subsamples confirms the role of the first principal component, but highlight that its contribution to the overall risk reduces in the second sample, 2013 - 2017, where the residual term increases and the second component has some role.

Finally, if we consider the regression of the latent factors on the set of economic drivers, we find a confirmation of the full sample outcome: while the principal component factors are more homogeneous in the exposure to economic drivers, the factors extracted by a multilevel model are heterogeneous in the exposures to exogenous world variables. This evidence holds in both subsamples.

#### 8 Concluding remarks

Latent factor models represent a common tool in the analysis of macroeconomic and financial variables. In many cases, to estimate factors we resort to principal component analyses. However, this approach has some limitations, in particular for the economic interpretation of the latent factors. We introduced a special multilevel model to overcome this limitation by decomposing a collection of variables into mutually exclusive groups. The model used the groups to identify a set of pervasive, semi-pervasive, and group-specific latent factors. These factors turned out to be easier to interpret, thanks to their relation with known groups of variables. Further, the use of the multilevel model might provide an easier interpretation to the loadings and the identification of risk sources. We supported the model by simulations as well as by an empirical analyses based on global CDS spreads. The approach we put forward might be of interest in all areas in which the variables of interest can be easily classified according to extra-sample information.

#### References

- Alessi, L., Barigozzi, M., and Capasso, M. (2010). Improved penalization for determining the number of factors in approximate factor models. *Statistics & Probability Letters*, 80(23):1806–1813.
- Ang, A. and Longstaff, F. (2013). Systemic sovereign credit risk: Lessons from the u.s. and europe. *Journal of Monetary Economics*, 60(5):493–510.
- Augustin, P. and Tedongap, R. (2016). Real economic shocks and sovereign credit risk. *Journal of Financial and Quantitative Analysis*, 51(2):541–587.
- Bai, J. (2003). Inferential theory for factor models of large dimensions. *Econometrica*, pages 135–171.
- Bai, J. and Ng, S. (2002). Determining the number of factors in approximate factor models. *Econometrica*, 70(1):191–221.
- Boivin, J. and Ng, S. (2006). Are more data always better for factor analysis? *Journal of Econometrics*, 132(1):169–194.
- Breitung, J. and Eickmeier, S. (2016). Analyzing international business and financial cycles using multi-level factor models: A comparison of alternative approaches. Dynamic Factor Models (Advances in Econometrics, Volume 35) Emerald Group Publishing Limited, 35:177–214.
- Caporin, M., Pelizzon, L., and Plazzi, A. (2017). Does monetary policy impact international market co-movements? *Swiss Finance Institute Research Papers Series*, 17-47.
- Caporin, M., Pelizzon, L., Ravazzolo, F., and Rigobon, R. (2018). Measuring sovereign contagion in europe. *Journal of Financial Stability*, 34:150–181.
- Choi, I., Kim, D., Kim, Y. J., and Kwark, N.-S. (2018). A multilevel factor model: Identification, asymptotic theory and applications. *Journal of Applied Econometrics*, 33(3):355–377.
- Fabozzi, A., Giacometti, R., and Tsuchida, N. (2016). Factor decomposition of the eurozone sovereign cds spreads. *Journal of International Money and Finance*, 65:1–23.
- Fender, I., Hayo, B., and Neuenkirch, M. (2012). Daily pricing of emerging market sovereign cds before and during the global financial crisis. *Journal of Banking* and Finance, 36(10):2786–2794.
- Forni, M., Hallin, M., Lippi, M., and Reichlin, L. (2000). The generalized dynamic-factor model: Identification and estimation. *The Review of Economics* and Statistics, 82(4):540–554.

- Forni, M., Hallin, M., Lippi, M., and Reichlin, L. (2004). The generalized dynamic factor model consistency and rates. *Journal of Econometrics*, 119(2):231–255.
- Forni, M., Hallin, M., Lippi, M., and Reichlin, L. (2005). The generalized dynamic factor model: one-sided estimation and forecasting. *Journal of the American Statistical Association*, 100(471):830–840.
- Kocsis, Z. and Monostori, Z. (2016). The role of country-specific fundamentals in sovereing cds spreas: Eastern european experiences. *Emerging Markets Review*, 27:140–168.
- Kose, M. A., Otrok, C., and Whiteman, C. H. (2003). International business cycles: World, region, and country-specific factors. *American Economic Review*, 93(4):1216–1239.
- Longstaff, F., Pan, J., Pedersen, L., and Singleton, K. (2011). How sovereing is sovereign credit risk? *American Economic Journal: Macroeconomics*, 3(April):75–103.
- Meucci, A. (2007). Risk contributions from generic user-defined factors. *Risk*, June:84–88.
- Rodríguez-Caballero, C. V. and Ergemen, Y. E. (2017). Estimation of a dynamic multilevel factor model with possible long-range dependence. Technical report, Universidad Carlos III de Madrid. Departamento de Estadística.
- Roncalli, T. and Weisang, G. (2016). Risk parity portfolios with risk factors. *Quantitative Finance*, 16(3):377–388.
- Stock, J. H. and Watson, M. W. (2002). Forecasting using principal components from a large number of predictors. *Journal of the American Statistical Association*, 97(460):1167–1179.
- Wang, P. (2010). Large dimensional factor models with a multi-level factor structure: Identification, estimation, and inference. Working paper. Hong Kong University of Science and Technology.

## Appendix

Factor/PC	PC1	PC2	PC3	PC4
Global	0.879	0.408	0.197	0.143
F12	0.205	-0.462	0.061	0.009
F13	-0.090	-0.088	0.163	0.507
F23	0.199	-0.012	-0.958	0.126
F1	0.055	-0.113	0.005	-0.040
F2	0.070	0.151	-0.026	-0.776
F3	0.360	-0.759	0.059	-0.140

### A Subsample 2009 - 2013

Table A1: Correlation between principal components (on columns) and factors of a multilevel model based on Debt/GDP ratio (1 is High, 2 is Medium and 3 is Low) in the subsample 2009 - 2013.



Figure A1: Box plots of estimated coefficients (loadings) for country groups based on the Debt/GDP ratio (High, Medium and Low levels) in the subsample 2009 - 2013. By row, from top to bottom, left to right: loadings to the global factor; loadings to the HM factor; loadings to the ML factor; loadings to the HL factor; loadings to the single group factor.



Figure A2: Box plots of estimated coefficients (loadings) for country groups based on the sovereing debt rating (High, High-Medium, Medium-Low, and Low levels) in the subsample 2009 - 2013. By row, from top to bottom, left to right loadings to the factors: global; H-HM-ML, H-ML-L, HM-ML-L; H-HM; H-ML; H-L; HM-ML; HM-L; ML-L; single.



Figure A3: Box plots of loadings to the principal components from 1 to 4 in the subsample 2009 - 2013.



Figure A4: Box plots of estimated coefficients (loadings) to the principal components for country groups based on the Debt/GDP ratio (High, Medium and Low levels) in the subsample 2009 - 2013. By row, from top to bottom, left to right: loadings to the first factor; loadings to the second factor; loadings to the third factor; loadings to the fourth factor.



Figure A5: Box plots of estimated coefficients (loadings) to the principal components for country groups based on the sovereign debt rating (High, High-Medium, Medium-Low, and Low levels) in the subsample 2009 - 2013. By row, from top to bottom, left to right: loadings to the first factor; loadings to the second factor; loadings to the third factor; loadings to the fourth factor.



Figure A6: Box plots of the fraction of variance explained by the various methods in the subsample 2009 - 2013. Top row, left plot, multilevel factor model with country rating groups, right plot, multilevel factor model with Debt/GDP groups. Bottom row, PCA model evaluated on the same groups of the top plots.



Figure A7: Box plots of the correlations between the original series and the residuals from the multilevel models and the PCA model in the subsample 2009 - 2013.

Factor/PC	PC1	PC2	PC3	PC4
GLOBAL	0.852	0.306	-0.276	-0.089
F123	0.344	-0.151	0.712	0.219
F124	0.130	0.127	0.329	-0.661
F134	-0.219	0.173	0.180	-0.288
F234	-0.025	0.479	0.430	0.290
F12	0.025	0.028	0.053	0.003
F13	-0.038	-0.050	0.021	0.198
F13	0.007	-0.079	0.078	0.183
F23	-0.047	0.050	0.054	-0.039
F24	0.026	-0.016	-0.084	-0.103
F34	-0.072	-0.240	0.176	-0.279
F1	0.227	-0.699	0.053	-0.023
F2	0.142	-0.170	-0.078	0.052
F3	-0.009	0.091	-0.101	0.238
F4	-0.093	0.085	-0.118	-0.182

Table A2: Correlation between principal components (on columns) and factors of a multilevel model based on country rating (1 is High, 2 is High-Medium, 3 is Medium-Low, and 4 is Low) in the subsample 2009 - 2013.

	<u> </u>	*** * * * *	*** * *				
	Global	High-Med	High-Low	Med-Low	High	Medium	Low
Global	0.802	0.030	-0.285	0.406	-0.202	0.154	0.098
H-HM-ML	0.412	0.306	0.071	-0.626	0.181	-0.245	0.160
H-HM-L	0.144	0.049	-0.187	-0.336	0.286	0.535	-0.026
H-ML-L	-0.142	-0.039	-0.237	-0.279	-0.279	0.246	-0.146
HM-ML-L	0.300	-0.441	0.389	-0.345	-0.137	-0.027	-0.232
H-HM	0.041	-0.018	-0.084	-0.036	0.248	-0.074	-0.045
H-ML	-0.020	0.259	-0.256	-0.038	0.069	-0.447	-0.172
H-L	0.012	-0.091	-0.291	-0.112	-0.134	-0.438	0.135
HM-ML	-0.014	0.025	-0.067	-0.049	-0.017	-0.044	-0.088
HM-L	-0.022	0.170	-0.148	0.035	0.334	0.135	-0.105
ML-L	-0.175	0.084	-0.232	-0.235	-0.410	0.129	0.289
Н	-0.078	0.233	0.247	0.041	0.049	0.066	0.708
HM	0.040	-0.031	-0.003	0.126	0.223	-0.064	0.225
ML	0.045	-0.026	0.109	0.119	0.051	-0.161	-0.138
L	-0.110	-0.143	-0.347	0.031	0.242	0.110	-0.070

Table A3: Correlation between the factors of the two multilevel models in the subsample 2009 - 2013. On columns the factors based on the Debt/GDP classification (High, Medium and Low) while on the rows the factors based on the country rating classification (High, Medium-High, Medium-Low and Low).

	High	Med-High	Med-Low	Low
GLOBAL	42.91	64.37	71.15	68.86
F123	17.91	14.61	12.43	0
F124	3.61	7.07	0	2.77
F134	2.86	0	8.53	2.24
F234	0	0.38	0.92	-0.01
F12	0.37	2.83	0	0
F13	1.36	0	0.58	0
F14	0.03	0	0	0.03
F23	0	3.49	0.82	0
F24	0	-0.07	0	0.61
F34	0	0	3.3	14.56
F1	30.03	0	0	0
F2	0	3.97	0	0
F3	0	0	0.26	0
F4	0	0	0	3.1
Residual	0.91	3.35	2.02	7.85

Table A4: Risk contribution of the latent factor to the total risk of equally weighted portfolios by country groups based on ratings in the subsample 2009 - 2013.

	High	Med	Low
GLOBAL	45.18	84.45	80.2
F12	45.75	4.14	0
F13	1.61	0	1.66
F23	0	3.85	5.87
F1	4.22	0	0
F2	0	3.51	0
F3	0	0	9.74
Residual	3.24	4.05	2.52

Table A5: Risk contribution of the latent factor to the total risk of equally weighted portfolios by country groups based on Debt/GDP in the subsample 2009 - 2013.

		Debt/GDP			Rat	ing	
	High	Medium	Low	High	Med-High	Med-Low	Low
PC1	83.27	88.59	94.77	90.16	80.83	91.75	89.01
PC2	4.68	4.34	0.30	0.00	4.15	0.06	0.93
PC3	0.99	0.13	0.15	0.06	7.58	0.87	0.00
PC4	2.17	0.02	0.59	1.15	0.42	0.49	0.36
Residual	8.89	6.91	4.19	8.62	7.01	6.82	9.70

Table A6: Risk contribution of the PCA model on the total risk of equally weighted portfolios by country groups based on Debt/GDP or on country rating in the sub-sample 2009 - 2013.

2013
2009 - 2013
osample
n the sul
group ii
0 GDP g
bt t
lts for De
n Resu
tegression Resul
Table A7: R
Tal

			Depe	Dependent variable:	:•		
-	Global	F12	F13	F23	F1	F2	F3
	(1)	(2)	(3)	(4)	(5)	(9)	(2)
XEquity_D	57.316**	-12.176	-2.410	-53.054***	-0.212	3.348	-8.453***
	(22.829)	(10.092)	(11.226)	(12.213)	(0.257)	(3.366)	(2.381)
XEquity_E	-190.510***	15.846**	1.963	47.290***	-0.105	-0.096	3.118*
	(15.968)	(6.900)	(8.345)	(8.825)	(0.244)	(1.833)	(1.711)
XVIX	3.821 (2.744)	-0.619 (1.099)	-1.495 (1.028)	4.934*** (1.279)	0.008 (0.035)	0.547 (0.415)	-0.481* (0.284)
XOil	13.249*** (5.046)	0.917 (2.241)	-0.567 (2.400)	-11.153*** (3.257)	0.042 (0.075)	0.424 (0.663)	$1.438^{**}$ (0.651)
XUscurr	228.226***	5.962	21.474	192.016***	-1.558	-0.026	-13.452*
	(61.705)	(28.364)	(33.628)	(34.378)	(1.145)	(7.638)	(7.119)
XBond_W	69.615*	4.695	-3.410	106.645***	-0.866	-9.421*	-20.060***
	(41.646)	(19.895)	(18.693)	(24.998)	(0.929)	(5.303)	(5.082)
XBond_W_10+_minus_1_3	9.642	-38.129*	25.361	-38.825	0.371	4.201	12.699**
	(47.729)	(21.136)	(17.791)	(27.143)	(0.811)	(4.706)	(5.796)
XBond_W_A_minus_AAA	-103.860***	-85.119***	16.012	12.395	-1.364***	-1.512	-23.725***
	(33.597)	(14.905)	(14.109)	(14.323)	(0.496)	(2.558)	(3.606)
XBond_W_C_BBB_minus_AAA	-93.455**	26.057	-17.548	-23.390	$1.541^{***}$	-3.975	-5.410
	(44.524)	(19.191)	(21.771)	(20.554)	(0.543)	(4.167)	(5.043)
Constant	0.111 (0.090)	-0.018 (0.044)	0.004 (0.040)	0.015 (0.052)	-0.001 (0.001)	0.001 (0.011)	-0.001 (0.012)
Adjusted R <sup>2</sup>	0.480	0.064	0.002	0.280	0.017	0.003	0.123

2013
2009 -
ole 2
samj
sqns
the
p in
intries group in the subsa
tries
nc
Rating co
Ra
s for
sults
Re
ession R
gres
: Re
A8:
ble
Tabl

							Depend	Dependent variable:							
	Global	F123	F124	F134	F234	F12	F13	F14	F23		F24	F34	F1	F2	F3
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)	(13)	(14)	(15)
XEquity_D	10.724	373.289***	8.820	$19.842^{*}$	25.424***	-9.406*	1.589	$-16.414^{**}$	2.608	2.608	-4.336	-1.601***	-861.583**	-46.192**	4,367.799*
	(23.409)	(113.087)	(9.955)	(10.847)	(9.680)	(5.523)	(6.637)	(6.845)	(5.791)	(5.791)	(10.882)	(0.521)	(351.957)	(23.503)	(2,406.705)
XEquity_E	-116.205***	-556.129***	-12.973**	-9.282	-41.592***	2.886	-2.738	3.114	-5.446	-5.446	13.746*	0.675*	495.398*	14.447	2,141.050
	(17.126)	(86.032)	(6.458)	(7.488)	(6.794)	(3.663)	(4.643)	(4.942)	(4.058)	(4.058)	(7.970)	(0.394)	(260.357)	(17.530)	(1,888.803)
XIVX	8.862***	-26.344*	-0.634	0.563	-0.626	-0.679	-0.433	-1.257*	-0.241	-0.241	-1.664	-0.070	12.932	-2.452	685.384**
	(2.629)	(15.708)	(1.174)	(1.140)	(1.121)	(0.616)	(0.738)	(0.739)	(0.621)	(0.621)	(1.104)	(0.066)	(35.997)	(2.306)	(275.876)
XOil	3.202	$104.930^{***}$	3.803*	3.175	1.560	3.389**	-1.317	0.771	0.774	0.774	2.416	0.203	111.051	7.630	238.946
	(4.552)	(34.013)	(2.229)	(2.505)	(2.128)	(1.476)	(1.423)	(1.779)	(1.253)	(1.253)	(2.749)	(0.135)	(95.415)	(5.467)	(657.967)
XUscurr	299.882***	-739.656**	-26.792	-95.007***	-18.677	17.161	-6.221	-17.338	-12.655	-12.655	-39.602	-3.518**	447.687	83.136	1,773.222
	(68.368)	(330.760)	(26.499)	(31.761)	(28.295)	(16.029)	(16.948)	(20.703)	(14.979)	(14.979)	(35.186)	(1.544)	(1,018.604)	(67.519)	( $8,451.830$ )
XBond_W	101.551**	-610.858**	-25.991	-20.044	-8.994	35.061***	8.449	10.527	7.653	7.653	-38.301	-3.767***	1,160.460*	127.085***	-6,508.396
	(44.580)	(242.791)	(22.571)	(21.990)	(19.433)	(11.313)	(12.858)	(14.246)	(10.650)	(10.650)	(23.354)	(1.057)	(696.487)	(47.688)	(6,139.603)
XBond_W_10+_minus_1_3	-18.214	260.945	6.807	14.081	29.656	-37.228***	-9.692	-12.219	8.908	8.908	2.699	2.652**	118.574	-26.962	7,300.757
	(44.047)	(278.591)	(20.873)	(20.406)	(19.596)	(11.275)	(11.657)	(13.711)	(13.233)	(13.233)	(23.938)	(1.145)	(713.507)	(52.374)	(6,188.734)
XBond_W_A_minus_AAA	-84.519***	-1,015.989***	-21.871**	23.822***	80.163***	-3.611	-0.883	3.589	9.617*	9.617*	9.587	-2.932***	-891.265**	74.959***	2,373.909
	(32.303)	(234.830)	(8.734)	(8.875)	(16.381)	(8.262)	(7.791)	(8.303)	(5.658)	(5.658)	(13.568)	(1.024)	(353.549)	(21.678)	(2,731.301)
XBond_W_C_BBB_minus_AAA	-105.667**	279.285	5.392	27.905*	-43.024**	-9.762	$19.203^{*}$	3.538	-4.461	-4.461	-29.031	-0.755	6.136	22.479	2,083.162
	(45.325)	(221.444)	(12.614)	(16.571)	(20.256)	(8.488)	(10.547)	(12.607)	(11.511)	(11.511)	(19.443)	(1.122)	(661.860)	(41.904)	(5,165.173)
Constant	0.103 (0.089)	-0.085 (0.526)	-0.002 (0.041)	-0.015 (0.043)	0.031 (0.041)	0.003 (0.026)	-0.004 (0.026)	0.001 (0.030)	0.001 (0.026)	0.001 (0.026)	0.002 (0.047)	-0.0002 (0.002)	-0.302 (1.525)	-0.005 (0.103)	-2.891 (12.155)
Adjusted R <sup>2</sup>	0.473	0.126	0.009	0110	0 117	0.011	0.00003	-0.004	0.008	0.000	0.045	0.057	0.031	0.014	0.011

		Dependent	variable:	
	F1	F2	F3	F4
	(1)	(2)	(3)	(4)
 XEquity_D	-0.785	7.672***	6.742***	-1.620
	(2.998)	(2.004)	(1.638)	(1.622)
XEquity_E	-19.692***	-8.388***	-7.990***	-0.581
	(2.156)	(1.273)	(1.192)	(1.008)
XVIX	0.632*	0.483**	-0.679***	-0.219
	(0.348)	(0.231)	(0.174)	(0.176)
XOil	2.038***	-0.401	1.640***	-0.384
	(0.751)	(0.411)	(0.466)	(0.332)
XUscurr	33.510***	13.058***	-21.120***	5.286
	(8.834)	(5.046)	(4.806)	(4.116)
XBond_W	3.883	11.320***	-13.013***	6.099**
	(5.696)	(3.469)	(3.645)	(2.743)
XBond_W_10+_minus_1_3	2.148	-3.062	6.139	-1.139
	(6.579)	(3.560)	(3.951)	(2.592)
XBond_W_A_minus_AAA	-34.798***	16.930***	-3.982**	3.095**
	(4.898)	(2.300)	(1.939)	(1.337)
XBond W C BBB minus AAA	-13.197**	-3.040	0.684	-0.670
	(6.152)	(3.794)	(2.888)	(2.437)
Constant	0.013	0.005	-0.0005	0.001
	(0.012)	(0.008)	(0.007)	(0.006)
$\Delta$ diusted $\mathbf{P}^2$	0.570	0.147	0.196	0.025
Adjusted R <sup>2</sup>	0.570	0.147	0.196	0.025

Table A9: Regression Results for PCA in the subsample 2009 - 2013

## B subsample 2013 - 2017

Factor/PC	PC1	PC2	PC3	PC4
Global	0.864	0.153	0.388	0.275
F12	0.185	0.334	-0.455	-0.122
F13	0.141	0.012	0.127	-0.642
F23	-0.327	0.853	0.360	0.072
F1	0.178	0.061	-0.432	0.164
F2	0.131	-0.026	0.234	-0.631
F3	0.208	0.363	-0.505	-0.226

Table B1: Correlation between principal components (on columns) and factors of a multilevel model based on Debt/GDP ratio (1 is High, 2 is Medium and 3 is Low) in the subsample 2013 - 2017.



Figure B1: Box plots of estimated coefficients (loadings) for country groups based on the Debt/GDP ratio (High, Medium and Low levels) in the subsample 2013 - 2017. By row, from top to bottom, left to right: loadings to the global factor; loadings to the HM factor; loadings to the ML factor; loadings to the HL factor; loadings to the single group factor.



Figure B2: Box plots of estimated coefficients (loadings) for country groups based on the sovereing debt rating (High, High-Medium, Medium-Low, and Low levels) in the subsample 2013 - 2017. By row, from top to bottom, left to right loadings to the factors: global; H-HM-ML, H-ML-L, HM-ML-L; H-HM; H-ML; H-L; HM-ML; HM-L; ML-L; single.



Figure B3: Box plots of loadings to the principal components from 1 to 4 in the subsample 2013 - 2017.



Figure B4: Box plots of estimated coefficients (loadings) to the principal components for country groups based on the Debt/GDP ratio (High, Medium and Low levels) in the subsample 2013 - 2017. By row, from top to bottom, left to right: loadings to the first factor; loadings to the second factor; loadings to the third factor; loadings to the fourth factor.



Figure B5: Box plots of estimated coefficients (loadings) to the principal components for country groups based on the sovereign debt rating (High, High-Medium, Medium-Low, and Low levels) in the subsample 2013 - 2017. By row, from top to bottom, left to right: loadings to the first factor; loadings to the second factor; loadings to the third factor; loadings to the fourth factor.



Figure B6: Box plots of the fraction of variance explained by the various methods in the subsample 2013 - 2017. Top row, left plot, multilevel factor model with country rating groups, right plot, multilevel factor model with Debt/GDP groups. Bottom row, PCA model evaluated on the same groups of the top plots.



Figure B7: Box plots of the correlations between the original series and the residuals from the multilevel models and the PCA model in the subsample 2013 - 2017.

Factor/PC	PC1	PC2	PC3	PC4
GLOBAL	0.884	-0.057	0.302	0.085
F123	-0.118	-0.722	0.165	0.047
F124	-0.009	0.140	-0.061	0.551
F134	0.029	0.152	0.232	-0.204
F234	-0.192	0.519	0.574	0.014
F12	-0.121	-0.056	-0.167	0.292
F13	-0.008	0.040	-0.081	0.179
F13	0.074	-0.083	0.076	-0.007
F23	0.015	-0.044	0.147	-0.005
F24	-0.040	-0.012	-0.070	-0.125
F34	0.196	-0.084	0.000	0.127
F1	-0.226	-0.251	0.634	0.073
F2	-0.229	-0.134	0.145	0.182
F3	0.009	0.099	0.026	0.596
F4	0.051	-0.216	0.020	0.077

Table B2: Correlation between principal components (on columns) and factors of a multilevel model based on country rating (1 is High, 2 is High-Medium, 3 is Medium-Low, and 4 is Low) in the subsample 2013 - 2017.

GlobalHigh-MedHigh-LowMed-LowHighMediumLowGlobal0.904-0.0290.147-0.2260.0000.0520.028H-HM-ML-0.115-0.222-0.159-0.608-0.584-0.046-0.211H-HM-L0.1520.090-0.5200.1100.006-0.313-0.034H-ML-L0.096-0.0080.1510.158-0.2760.1060.062HM-ML-L0.160-0.054-0.0600.642-0.6030.0310.009H-HM-0.068-0.1740.060-0.046-0.008-0.6280.136H-ML0.0280.2610.119-0.022-0.148-0.411-0.116H-L0.077-0.459-0.056-0.0560.0460.1190.284HM-ML0.0470.034-0.1850.004-0.1080.305-0.120HM-L0.1010.2060.086-0.0340.0070.046-0.101ML-L0.1730.246-0.106-0.1000.2160.097-0.327H0.016-0.4670.3300.1800.101-0.062-0.724HM0.181-0.225-0.4050.0550.0810.167-0.224ML0.181-0.072-0.120-0.220-0.139-0.041-0.195								
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		Global	High-Med	High-Low	Med-Low	High	Medium	Low
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Global	0.904	-0.029	0.147	-0.226	0.000	0.052	0.028
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	H-HM-ML	-0.115	-0.222	-0.159	-0.608	-0.584	-0.046	-0.211
HM-ML-L0.160-0.054-0.0600.642-0.6030.0310.009H-HM-0.068-0.1740.060-0.046-0.008-0.6280.136H-ML0.0280.2610.119-0.022-0.148-0.411-0.116H-L0.077-0.459-0.056-0.0560.0460.1190.284HM-ML0.0470.034-0.1850.004-0.1080.305-0.120HM-L-0.1010.2060.086-0.0340.0070.046-0.101ML-L0.1730.246-0.106-0.1000.2160.097-0.327H0.016-0.4670.3300.1800.101-0.062-0.724HM-0.119-0.225-0.4050.0550.0810.167-0.224ML0.181-0.088-0.2770.1480.173-0.353-0.125	H-HM-L	0.152	0.090	-0.520	0.110	0.006	-0.313	-0.034
H-HM-0.068-0.1740.060-0.046-0.008-0.6280.136H-ML0.0280.2610.119-0.022-0.148-0.411-0.116H-L0.077-0.459-0.056-0.0560.0460.1190.284HM-ML0.0470.034-0.1850.004-0.1080.305-0.120HM-L-0.1010.2060.086-0.0340.0070.046-0.101ML-L0.1730.246-0.106-0.1000.2160.097-0.327H0.016-0.4670.3300.1800.101-0.062-0.724HM-0.119-0.225-0.4050.0550.0810.167-0.224ML0.181-0.088-0.2770.1480.173-0.353-0.125	H-ML-L	0.096	-0.008	0.151	0.158	-0.276	0.106	0.062
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	HM-ML-L	0.160	-0.054	-0.060	0.642	-0.603	0.031	0.009
H-L0.077-0.459-0.056-0.0560.0460.1190.284HM-ML0.0470.034-0.1850.004-0.1080.305-0.120HM-L-0.1010.2060.086-0.0340.0070.046-0.101ML-L0.1730.246-0.106-0.1000.2160.097-0.327H0.016-0.4670.3300.1800.101-0.062-0.724HM-0.119-0.225-0.4050.0550.0810.167-0.224ML0.181-0.088-0.2770.1480.173-0.353-0.125	H-HM	-0.068	-0.174	0.060	-0.046	-0.008	-0.628	0.136
HM-ML0.0470.034-0.1850.004-0.1080.305-0.120HM-L-0.1010.2060.086-0.0340.0070.046-0.101ML-L0.1730.246-0.106-0.1000.2160.097-0.327H0.016-0.4670.3300.1800.101-0.062-0.724HM-0.119-0.225-0.4050.0550.0810.167-0.224ML0.181-0.088-0.2770.1480.173-0.353-0.125	H-ML	0.028	0.261	0.119	-0.022	-0.148	-0.411	-0.116
HM-L-0.1010.2060.086-0.0340.0070.046-0.101ML-L0.1730.246-0.106-0.1000.2160.097-0.327H0.016-0.4670.3300.1800.101-0.062-0.724HM-0.119-0.225-0.4050.0550.0810.167-0.224ML0.181-0.088-0.2770.1480.173-0.353-0.125	H-L	0.077	-0.459	-0.056	-0.056	0.046	0.119	0.284
ML-L0.1730.246-0.106-0.1000.2160.097-0.327H0.016-0.4670.3300.1800.101-0.062-0.724HM-0.119-0.225-0.4050.0550.0810.167-0.224ML0.181-0.088-0.2770.1480.173-0.353-0.125	HM-ML	0.047	0.034	-0.185	0.004	-0.108	0.305	-0.120
H0.016-0.4670.3300.1800.101-0.062-0.724HM-0.119-0.225-0.4050.0550.0810.167-0.224ML0.181-0.088-0.2770.1480.173-0.353-0.125	HM-L	-0.101	0.206	0.086	-0.034	0.007	0.046	-0.101
HM-0.119-0.225-0.4050.0550.0810.167-0.224ML0.181-0.088-0.2770.1480.173-0.353-0.125	ML-L	0.173	0.246	-0.106	-0.100	0.216	0.097	-0.327
ML 0.181 -0.088 -0.277 0.148 0.173 -0.353 -0.125	Н	0.016	-0.467	0.330	0.180	0.101	-0.062	-0.724
	HM	-0.119	-0.225	-0.405	0.055	0.081	0.167	-0.224
L 0.051 -0.072 -0.120 -0.220 -0.139 -0.041 -0.019	ML	0.181	-0.088	-0.277	0.148	0.173	-0.353	-0.125
	L	0.051	-0.072	-0.120	-0.220	-0.139	-0.041	-0.019

Table B3: Correlation between the factors of the two multilevel models in the subsample 2013 - 2017. On columns the factors based on the Debt/GDP classification (High, Medium and Low) while on the rows the factors based on the country rating classification (High, Medium-High, Medium-Low and Low).

	High	Med-High	Med-Low	Low
GLOBAL	32	64.55	75.37	48.59
F123	9.83	3.57	0.91	0
F124	0	0.66	0	0.34
F134	1.75	0	0.01	0.37
F234	0	0.2	11.47	10.01
F12	2.36	5.9	0	0
F13	0.13	0	0.08	0
F14	-0.03	0	0	6.08
F23	0	0.25	0.56	0
F24	0	0.13	0	0.07
F34	0	0	8.87	9.42
F1	51.16	0	0	0
F2	0	23.04	0	0
F3	0	0	0.08	0
F4	0	0	0	22.58
Residual	2.79	1.71	2.66	2.53

Table B4: Risk contribution of the latent factor to the total risk of equally weighted portfolios by country groups based on ratings in the subsample 2013 - 2017.

	High	Med	Low
GLOBAL	28.92	71.31	67.04
F12	36.75	3.77	0
F13	6.25	0	3.65
F23	0	11.71	18.25
F1	17.98	0	0
F2	0	7.95	0
F3	0	0	7.05
Residual	10.1	5.26	4

Table B5: Risk contribution of the latent factor to the total risk of equally weighted portfolios by country groups based on Debt/GDP in the subsample 2013 - 2017.

		Debt/GDP			Rat	ing	
	High	Medium	Low	High	Med-High	Med-Low	Low
PC1	63.07	86.49	91.90	79.49	84.52	86.31	74.42
PC2	11.12	0.43	0.55	0.03	4.92	0.86	0.46
PC3	7.90	2.10	0.29	1.64	1.80	0.48	1.54
PC4	2.32	0.06	0.00	1.61	1.17	1.07	0.56
Residual	15.59	10.92	7.27	17.24	7.58	11.29	23.03

Table B6: Risk contribution of the PCA model on the total risk of equally weighted portfolios by country groups based on Debt/GDP or on country rating in the sub-sample 2013 - 2017.

			nepe	Dependent variable:	·.		
	Global	F12	F13	F23	F1	F2	F3
	(1)	(2)	(3)	(4)	(5)	(9)	(7)
XEquity_D	-73.627***	-41.776**	-27.111*	53.129***	-0.845***	0.128	-9.403**
	(24.300)	(21.283)	(15.330)	(19.266)	(0.297)	(2.679)	(4.300)
XEquity_E	-231.097***	26.308***	14.995*	-52.619***	0.227	0.490	4.065
	(14.872)	(8.470)	(8.220)	(11.751)	(0.164)	(1.515)	(2.626)
XVIX	-6.121*** (1.434)	-1.223 (1.389)	-0.895 (0.882)	-6.546*** (1.246)	0.026 (0.021)	0.034 (0.156)	-0.325 (0.226)
XOil	6.290	2.456	$3.583^{*}$	18.951***	-0.035	-0.731	0.734
	(3.851)	(2.097)	(2.091)	(2.908)	(0.042)	(0.501)	(0.752)
XUscurr	-89.599	19.237	-8.603	-307.396***	-0.865	2.027	-5.599
	(56.029)	(28.227)	(31.258)	(45.412)	(0.654)	(6.135)	(8.493)
XBond_W	8.912	-4.158	-3.966	-189.052***	0.708	4.366	-7.918
	(58.085)	(30.862)	(28.780)	(42.937)	(0.641)	(6.213)	(7.615)
XBond_W_10+_minus_1_3	-48.131	9.409	-29.399	174.677***	-1.091**	-9.050**	3.147
	(39.410)	(22.854)	(21.723)	(31.404)	(0.515)	(4.140)	(5.003)
XBond_W_A_minus_AAA	-104.748**	-13.656	-75.190***	44.726	-3.270***	-0.816	0.811
	(45.034)	(23.963)	(25.278)	(30.468)	(0.576)	(4.131)	(6.028)
XBond_W_C_BBB_minus_AAA	-81.358	-60.218**	-15.571	43.722	-0.054	-14.369**	-16.384*
	(69.713)	(29.455)	(27.493)	(39.031)	(0.564)	(5.763)	(8.535)
Constant	0.063 (0.073)	0.009 (0.042)	0.015 (0.039)	0.008 (0.053)	0.001 (0.001)	0.001 (0.008)	0.004 (0.011)
Adjusted R <sup>2</sup>	0.400	0.018	0.016	0.288	0.107	0.006	0.009

Table B7: Regression Results for Debt to GDP group in the subsample 2013 - 2017.

Table B8: Regression Results for Rating countries group in the subsample 2013 - 2017.

Global (1) XEquity_D -91.710*** (24.822) XEquity_E -190.674*** (14.865) XVIX -3.920***	F123 (2) 24.326	F124	F134	F234	F12	F13	F14	Ē		61	121	н Ц	6	F3
	(2) 24.326							F23	3	F24	104		77	•
	24.326	(3)	(4)	(5)	(9)	6	(8)	(6)	(10)	(11)	(12)	(13)	(14)	(15)
	(36.045)	-9.586 (10.268)	-9.257 (11.386)	58.176*** (14.946)	6.681 (11.333)	7.286 (9.910)	12.129 (12.349)	$19.619^{**}$ (9.037)	19.619** (9.037)	-27.767 (20.127)	$1.412^{**}$ (0.672)	844.930*** (317.100)	159.361 (186.315)	$5,401.373^{*}$ (3,090.357)
·	65.760*** (19.767)	$-16.100^{***}$ (6.213)	-4.453 (6.227)	-56.656*** (9.730)	3.336 (6.429)	-4.267 (5.387)	-14.984*** (5.558)	-16.663*** (5.004)	-16.663*** (5.004)	0.237 (11.282)	-0.982*** (0.333)	-185.655 (174.050)	-484.335*** (114.998)	$-4,802.410^{***}$ (1,638.518)
	$6.640^{***}$	-1.090	-1.551**	-4.978***	-0.175	0.325	0.407	0.418	0.418	-0.609	0.029	10.281	1.575	98.428
	(2.414)	(0.685)	(0.677)	(0.981)	(0.701)	(0.518)	(0.809)	(0.597)	(0.597)	(1.256)	(0.042)	(20.659)	(10.595)	(159.633)
XOil 1.751 (3.739)	-21.849***	-1.117	1.355	$10.454^{***}$	-0.886	1.708	-4.816***	-1.352	-1.352	-6.153**	-0.012	-7.319	66.482**	74.613
	(5.339)	(1.654)	(1.610)	(2.315)	(1.934)	(1.487)	(1.493)	(1.697)	(1.697)	(3.132)	(0.087)	(44.588)	(33.241)	(421.888)
XUscurr 12.685 3	363.096***	-31.462	-7.345	-133.457***	5.955	7.227	-26.726	9.332	9.332	67.712*	-2.114*	880.138	-544.148	9,638.664
(57.017)	(75.922)	(21.548)	(24.086)	(37.423)	(21.992)	(20.655)	(17.455)	(21.301)	(21.301)	(39.715)	(1.111)	(698.025)	(441.658)	(6,304.923)
XBond_W 57.638	169.605**	-8.477	4.500	-117.300***	-6.023	-14.968	-21.000	-7.706	-7.706	$102.853^{***}$	-0.627	895.992	-323.330	3,138.340
(62.317)	(72.828)	(21.765)	(22.056)	(36.240)	(23.835)	(19.069)	(19.070)	(19.612)	(19.612)	(38.632)	(1.105)	(648.507)	(425.764)	(5,735.325)
XBond_W_10+_minus_1_3 -143.788***	-79.156	32.843**	20.139	108.879***	-2.444	23.539*	5.346	4.486	4.486	-50.346*	0.108	-582.512	516.116*	-3,590.120
(40.171)	(53.748)	(16.523)	(16.704)	(26.424)	(17.371)	(13.435)	(13.755)	(14.137)	(14.137)	(27.617)	(0.838)	(459.309)	(304.337)	(3,920.404)
XBond_W_A_minus_AAA -155.694*** 2	232.003***	36.535*	45.931**	$114.840^{***}$	-2.512	6.208	-8.002	39.947***	39.947***	-46.538	-1.123	263.765	-107.769	3,668.861
(52.827)	(58.989)	(19.831)	(20.385)	(30.025)	(16.289)	(14.257)	(15.224)	(13.573)	(13.573)	(28.732)	(0.969)	(547.284)	(290.557)	(4,292.510)
XBond_W_C_BBB_minus_AAA -116.080	66.273	34.586	8.548	-5.960	21.401	17.929	-10.983	-9.878	-9.878	-17.732	2.946**	586.252	780.597*	-2,130.473
(71.272)	(65.527)	(23.279)	(22.505)	(32.784)	(24.215)	(16.652)	(20.438)	(19.363)	(19.363)	(37.250)	(1.180)	(638.217)	(404.680)	(5,450.691)
Constant 0.066 (0.074)	-0.062 (0.096)	0.003 (0.031)	0.001 (0.032)	-0.012 (0.045)	-0.004 (0.031)	-0.006 (0.024)	-0.001 (0.027)	-0.008 (0.026)	-0.008 (0.026)	0.005 ( $0.050$ )	-0.0001 (0.002)	-0.330 (0.878)	0.003 (0.528)	-2.370 (7.983)
Adjusted R <sup>2</sup> 0.397	0.089	0.016	0.008	0.206	0.001	-0.001	0.013	0.015	0.015	0.044	0.022	0.014	0.029	0.015

		Dependent	variable:	
	F1	F2	F3	F4
	(1)	(2)	(3)	(4)
XEquity_D	-22.373***	-1.000	8.967***	2.956
	(3.745)	(3.241)	(3.141)	(2.177)
XEquity_E	-22.376***	-5.885***	-13.385***	-7.118***
	(2.116)	(1.779)	(1.498)	(1.214)
XVIX	-0.462**	-0.952***	-0.372*	0.034
	(0.214)	(0.188)	(0.199)	(0.129)
XOil	-0.282	2.537***	0.587	0.094
	(0.510)	(0.411)	(0.399)	(0.319)
XUscurr	7.718	-35.253***	-12.383**	-4.357
	(8.356)	(6.491)	(5.745)	(4.571)
XBond_W	15.307*	-22.321***	-6.667	-1.025
_	(8.594)	(5.879)	(5.530)	(4.611)
XBond_W_10+_minus_1_3	-24.292***	19.368***	5.159	6.492*
	(5.802)	(4.011)	(4.123)	(3.358)
XBond_W_A_minus_AAA	-30.882***	0.756	9.168**	3.146
	(7.411)	(4.169)	(4.308)	(3.604)
XBond_W_C_BBB_minus_AAA	-26.360**	-4.212	6.087	10.771**
	(10.340)	(5.410)	(5.133)	(4.289)
Constant	0.012	0.004	-0.001	-0.001
	(0.011)	(0.007)	(0.007)	(0.006)
Adjusted R <sup>2</sup>	0.458	0.175	0.111	0.044

Table B9: Regression Results for PCA in the subsample 2013 - 2017.

# Research Papers 2018



- 2018-15: Changli He, Jian Kang, Timo Teräsvirta and Shuhua Zhang: The Shifting Seasonal Mean Autoregressive Model and Seasonality in the Central England Monthly Temperature Series, 1772-2016
- 2018-16: Ulrich Hounyo and Rasmus T. Varneskov: Inference for Local Distributions at High Sampling Frequencies: A Bootstrap Approach
- 2018-17: Søren Johansen and Morten Ørregaard Nielsen: Nonstationary cointegration in the fractionally cointegrated VAR model
- 2018-18: Giorgio Mirone: Cross-sectional noise reduction and more efficient estimation of Integrated Variance
- 2018-19: Kim Christensen, Martin Thyrsgaard and Bezirgen Veliyev: The realized empirical distribution function of stochastic variance with application to goodness-of-fit testing
- 2018-20: Ruijun Bu, Kaddour Hadri and Dennis Kristensen: Diffusion Copulas: Identification and Estimation
- 2018-21: Kim Christensen, Roel Oomen and Roberto Renò: The drift burst hypothesis
- 2018-22: Russell Davidson and Niels S. Grønborg: Time-varying parameters: New test tailored to applications in finance and macroeconomics
- 2018-23: Emilio Zanetti Chini: Forecasters' utility and forecast coherence
- 2018-24: Tom Engsted and Thomas Q. Pedersen: Disappearing money illusion
- 2018-25: Erik Christian Montes Schütte: In Search of a Job: Forecasting Employment Growth in the US using Google Trends
- 2018-26: Maxime Morariu-Patrichi and Mikko Pakkanen: State-dependent Hawkes processes and their application to limit order book modelling
- 2018-27: Tue Gørgens and Allan H. Würtz: Threshold regression with endogeneity for short panels
- 2018-28: Mark Podolskij, Bezirgen Veliyev and Nakahiro Yoshida: Edgeworth expansion for Euler approximation of continuous diffusion processes
- 2018-29: Isabel Casas, Jiti Gao and Shangyu Xie: Modelling Time-Varying Income Elasticities of Health Care Expenditure for the OECD
- 2018-30: Yukai Yang and Luc Bauwens: State-Space Models on the Stiefel Manifold with A New Approach to Nonlinear Filtering
- 2018-31: Stan Hurn, Nicholas Johnson, Annastiina Silvennoinen and Timo Teräsvirta: Transition from the Taylor rule to the zero lower bound
- 2018-32: Sebastian Ankargren, Måns Unosson and Yukai Yang: A mixed-frequency Bayesian vector autoregression with a steady-state prior
- 2018-33: Carlos Vladimir Rodríguez-Caballero and Massimiliano Caporin: A multilevel factor approach for the analysis of CDS commonality and risk contribution